





Republic of Turkey Ministry of Health

Prevention and control of noncommunicable diseases in Turkey

The case for investment



Prevention and control of noncommunicable diseases in Turkey

The case for investment

Prepared for Ministry of Health of Turkey by WHO Regional Office for Europe

Authors: Anna Kontsevaya Jill Farrington Mehmet Balcılar Toker Ergüder







Abstract

Noncommunicable diseases (NCDs) such as cancer, cardiovascular diseases, diabetes and chronic respiratory diseases and their risk factors are an increasing public health and development challenge in Turkey. This report provides evidence through three analyses that NCDs reduce economic output, and discusses potential options in response, outlining details of their relative returns on investment. An economic burden analysis shows that economic losses from NCDs are equivalent to 3.6% of gross domestic product. An intervention costing analysis provides an estimate of the funding required to implement a set of policy interventions for prevention and clinical interventions. A cost–benefit analysis compares these implementation costs with the estimated health gains and identifies which policy packages would give the greatest returns on investment.

Keywords

CHRONIC DISEASE - ECONOMICS CHRONIC DISEASE - PREVENTION AND CONTROL COST-BENEFIT ANALYSIS TURKEY

Address requests about publications of the WHO Regional Office for Europe to: Publications WHO Regional Office for Europe UN City, Marmorvej 51 DK-2100 Copenhagen Ø, Denmark Alternatively, complete an online request form for documentation, health information, or for permission to quote or translate, on the Regional Office website (http://www.euro.who.int/pubrequest).

© World Health Organization 2018

All rights reserved. The Regional Office for Europe of the World Health Organization welcomes requests for permission to reproduce or translate its publications, in part or in full.

The designations employed and the presentation of the material in this publication do not imply the expression of any opinion whatsoever on the part of the World Health Organization concerning the legal status of any country, territory, city or area or of its authorities, or concerning the delimitation of its frontiers or boundaries. Dotted lines on maps represent approximate border lines for which there may not yet be full agreement.

The mention of specific companies or of certain manufacturers' products does not imply that they are endorsed or recommended by the World Health Organization in preference to others of a similar nature that are not mentioned. Errors and omissions excepted, the names of proprietary products are distinguished by initial capital letters.

All reasonable precautions have been taken by the World Health Organization to verify the information contained in this publication. However, the published material is being distributed without warranty of any kind, either expressed or implied. The responsibility for the interpretation and use of the material lies with the reader. In no event shall the World Health Organization be liable for damages arising from its use. The views expressed by authors, editors, or expert groups do not necessarily represent the decisions or the stated policy of the World Health Organization.

Contents

Ac	Acknowledgements iv			
AŁ	breviations	. v		
Ex	ecutive summary	vi		
1.	Introduction	.1		
	Purpose of the economic analysis component of the case for investment	1		
2.	Situation analysis: NCDs and risk factors	.3		
	Tobacco use			
	Harmful alcohol use			
	Physical inactivity			
	Unhealthy diet (salt) Metabolic risk factors			
2				
3.	Policies and treatments to reduce the NCD burden Tobacco			
	Alcohol			
	Physical inactivity			
	Unhealthy diet	10		
	CVD and diabetes			
	Summary	14		
4.	Methods			
	Calculation of economic burden of NCDs			
	Calculation of policy and interventions costs ROI 17	16		
	Resource utilization for ACSCs	17		
5.	Results	18		
	Economic burden assessment			
	Intervention cost assessment			
	Health benefit assessment Economic benefit assessment			
	ROI assessment			
	Hospitalizations and ambulatory care analysis			
6.	Conclusion	27		
Re	ferences	28		
Ar	nex 1. Contributors to the report	32		
Ar	nex 2. Data sources used for the economic burden analysis	33		
Δr	nex 3. Population projections: detailed methodology and results	33		
-41	Table A3. Population projections.			
Δr	inex 4. Data used for calculations of NCD burden			
	Table A4.1. Share of major NCDs in total health care expenditure			
	Table A4.1. Productivity data			

Acknowledgements

The authors would like to express their sincere gratitude to the Ministry of Health of Turkey and the national team who supported the data collection and analysis.

Economic analysis was carried out by Anna Kontsevaya, National Research Centre for Prevention, Moscow, Russian Federation, with contributions from Mehmet Balcilar, Health Economist; both were working as WHO consultants. This report was written by them with Jill Farrington, WHO Regional Office for Europe, and Toker Ergüder, WHO Country Office in Turkey.

The full list of experts who contributed to this report is given in Annex 1. The specific contributions of Emine Alp Meşe, Deputy Health Minister, Elif Güler Kazancı, Deputy Undersecretary, Hüseyin İlter, Acting Director-General of Public Health, İrfan Şencan, former Director of Public Health, Bekir Keskinkılıç, Deputy Director-General of Public Health, Banu Ekinci, Head of Department for Chronic Diseases and Elderly Health, and Zübeyde Özkan Altunay, Gülay Sarıoğlu, Ahmet Arıkan and Asiye Kapusuz Navruz, Turkish Directorate-General of Public Health and Tufan Nayır, Advisor to Deputy Health Minister in convening the national expert group and workshops and providing national materials to inform this report are particularly acknowledged.

Comments received from peer reviewer Dan Chisholm, WHO Regional Office for Europe, and Roy Small and Doug Webb of the United Nations Development Programme are much appreciated. The report also benefited from the advice of Melanie Bertram, WHO headquarters, Alexey Kulikov, United Nations Interagency Task Force on the Prevention and Control of Noncommunicable Diseases, and Dudley Tarlton, United Nations Development Programme, as well as insights from the approach taken for the noncommunicable disease investment cases for Viet Nam, Mongolia and Kyrgyzstan.

Thanks are also extended to Lydia Wanstall for language editing, Gül Menet and Sıla Saadet Toker for translation of the Turkish language version and Lars Møller for layout and typesetting the report, as well as Jesus Chew Ambrosio for making the arrangements.

The assessment was conducted under the overall guidance of Pavel Ursu, WHO Representative and Head of the WHO Country Office in Turkey, Gauden Galea and Bente Mikkelsen, former and current Directors of the Division of Noncommunicable Diseases and Promoting Health through the Life-course at the WHO Regional Office for Europe, and Nick Banatvala, Head of the Secretariat of the United Nations Interagency Task Force on the Prevention and Control of Noncommunicable Diseases.

Preparation of this report was coordinated by the WHO Regional Office for Europe and WHO Country Office in Turkey through the biennial collaborative agreement covering 2016/2017 between the Ministry of Health of Turkey and WHO. The report was co-financed through a voluntary contribution of the Government of the Russian Federation.

Abbreviations

ACSC	ambulatory care-sensitive condition
AMI	acute myocardial infarction
COPD	chronic obstructive pulmonary disease
CVD	cardiovascular disease
FCTC	Framework Convention on Tobacco Control [of WHO]
FMC	family medicine centre
GDP	gross domestic product
ICD-10	International Classification of Disease, 10th revision
IHD	ischaemic heart disease
MPOWER	monitor tobacco use and prevention policies; protect people from tobacco smoke; offer help to quit tobacco use; warn people about the dangers of tobacco; enforce bans on tobacco advertising, promotion and sponsorship; raise taxes on tobacco [WHO package]
NCD	noncommunicable disease
ROI	return on investment
SHAKE	surveillance; harness industry; adopt standards for labelling and marketing; knowledge; environment [WHO package]
SSI	Social Security Institute
STEPS	STEPwise approach to surveillance [of WHO]
TL	Turkish lira

Executive summary

Noncommunicable diseases (NCDs) such as cancer, cardiovascular disease (CVD), diabetes and chronic respiratory disease and their risk factors (tobacco use, harmful use of alcohol, unhealthy diet and physical inactivity) are an increasing public health and development challenge globally. In Turkey NCDs have a significant impact on the productivity of the workforce, causing disability and premature deaths; this has a negative impact on socioeconomic development. NCDs are responsible for 87.5% of all deaths in the country. The probability of premature death (death before the age of 70 years) from one of the four major NCDs for a person living in Turkey was around one in six (16.8%) in 2015.

The government was estimated to have spent 24.5 billion Turkish lira (TL) in 2015 on treatment of NCDs. A share of these funds could be redirected towards NCD prevention through implementation of national policies and more active primary care interventions, which are shown to be effective at boosting the economy and improving long-term health outcomes.

This report provides evidence that NCDs reduce economic output and discusses potential options in response, arriving at a set of interventions that are most suited to Turkey. Three analyses – each of which drew on the United Nations Interagency Working Group on Costing's strategic planning OneHealth Tool – were performed to arrive at these results.

- An economic burden analysis showed the scale of disruption to the economy from NCDs through assessment of their direct and indirect costs. Direct costs include government (public) health care costs for treating CVD, diabetes, cancer and respiratory disease. Indirect costs are based on disability payments, costs of absenteeism, costs of presenteeism and economic losses due to premature deaths among people of working age.
- An intervention costing analysis provided an estimate of the funding required to implement a set of NCD interventions. Costs of policy interventions were assessed using a WHO costing tool; costs for clinical interventions for CVD and diabetes were estimated using the WHO OneHealth Tool on the basis of prices of medications and supplies, as well as medical staff salaries.
- A return on investment (ROI) analysis compared the estimated implementation costs during the costing analysis with the estimated health gains and economic returns of a set of interventions. It measured the benefits that would be obtained for each sum spent on health intervention; for example, spending on salt reduction. As a preliminary step, the health benefits of these interventions were estimated using the OneHealth Tool and converted to monetary units.

In addition, a health care resources utilization analysis was performed on the ambulatory care-sensitive conditions (ACSCs) among the four main NCDs of interest.

The economic burden analysis found that government expenditure on health care for NCDs is just the tip of the iceberg: the hidden additional costs from lost productivity are higher than the health care costs. Altogether, the economic cost of NCDs to the Turkish economy in 2016 was TL 69.7 billion, which is equivalent to 3.6% of the country's annual gross domestic product.

Actions to prevent NCDs are relatively cheap and cost-effective. Their implementation requires engagement from sectors beyond health, such as finance, economy and agriculture, and benefits from the investments would accrue across the whole of government and society. The intervention costing analysis reviewed five packages of interventions for the prevention and control of NCDs which were considered priorities during the consultations with local experts: policy interventions for tobacco and alcohol use, salt intake and physical inactivity , and clinical interventions for CVD and diabetes in primary care. Policy packages to reduce consumption of tobacco, alcohol and salt and to increase physical activity were estimated to cost TL 384 million, TL 90.6 million,

TL 46.1 million and TL 124.3 million at five years, respectively. The CVD and diabetes interventions were found to be the most expensive options, costing TL 1619.0 million.

The ROI analysis was carried out for the same five packages of interventions. The results were generated over two scale-up periods: an initial five-year and a 15-year period. The analysis showed that the economic benefits of the interventions exceeded their implementation costs in both the shorter and longer time periods.

The most cost-effective interventions in Turkey were shown to be those for salt: the economic benefits of these packages far exceed their costs in the short and long term. The salt policy package achieved ROIs of 51 and 88 TL at five and 15 years, respectively. Reducing tobacco consumption – for example, through increased taxation – and increasing physical activity in the population would also be very cost-effective (with ROIs of 5.0 and 2.3 at 15 years, respectively), as would CVD and diabetes interventions in primary care – these achieved an ROI of 4.3 at 15 years.

Analysis of the ACSCs revealed that 942 047 hospitalizations occur annually for such conditions among the four main NCDs of interest (this figure represents those paid from social security funds; the number of hospitalizations for private clinics is unknown). This rate of hospitalizations relative to the number of registered patients is comparable to rates in other countries where such analysis has been performed. The analysis found signs of underuse of primary care among patients with some ACSCs.



1. Introduction

In 2012 noncommunicable diseases (NCDs) accounted for 87.5% of all deaths among people aged 30–70 years in Turkey – of these, 36.6% were due to cardiovascular disease (CVD), 38.9% to cancer, 7% to chronic respiratory diseases and 5.8% to diabetes (Turkish Statistical Institute, 2013). The latest WHO figures from 2015 show that a Turkish citizen has about a one-in-six chance (16.8%) of dying prematurely – that is, before the age of 70 years – from one of these four main NCDs, with a higher probability for men (22.5%) than women (11.6%) (WHO, 2017a). This highlights a significant opportunity to make progress on United Nations Sustainable Development Goal target 3.4, which aims to reduce premature mortality from NCDs by one third by 2030.

The impact of NCDs on human health is clear, but it is only one part of the story. NCDs also result in high economic costs, reaching far beyond direct health care costs. NCDs reduce productivity at a macro-economic level through interruption of full participation in the labour force and the subsequent impacts on individuals, their carers and the state. When individuals die prematurely, the labour output they would have produced in their remaining working years is lost. In addition, individuals who suffer from a disease are more likely to miss days of work (absenteeism) or to work at a reduced capacity while at work (presenteeism¹). In low- and middle-income countries it is estimated that, between 2011 and 2030, NCDs will cause more than US\$ 21 trillion in lost economic output, with nearly one third of that figure attributable to CVD alone (Bloom et al., 2011). For individuals and governments, spending on health can mean significant opportunity costs,² including decreased investment in education, transportation projects or other forms of human or physical capital that can produce long-term returns.

High human and economic costs highlight the need to reduce the burden of NCDs in Turkey. WHO recognizes that the risk of NCDs can be reduced by modifying four behaviours (tobacco use, harmful use of alcohol, an unhealthy diet and physical inactivity) and metabolic risk factors such as high blood pressure and cholesterol (WHO, 2013). Fig. 1 illustrates the determinants and risk factors that drive the development of NCDs, many of which are beyond the control of the health sector alone.

WHO developed a menu of policy options and cost-effective interventions to assist Member States to reduce the NCD burden within its global action plan for the prevention and control of NCDs 2013–2020 (WHO, 2013). These were recently updated at the World Health Assembly (WHO, 2017b) and include measures to reduce behavioural and metabolic risk factors for NCDs, as well as clinical interventions to prevent and treat disease. Given the high proportion of deaths caused by heart disease, stroke, myocardial infarction and other circulatory diseases in Turkey (estimated to be 47% in 2014 (WHO, 2014a)), the economic analysis detailed in this study focuses primarily on interventions that can reduce this burden of CVD.

Purpose of the economic analysis component of the case for investment

The negative economic impacts from NCDs are too often overlooked in budgetary allocation processes and in weighing the pros and cons of stronger fiscal and regulatory action. Quantifying the costs of NCD management and interventions to prevent and control NCDs, as well as their returns on investment (ROIs) in relation to the costs of inaction, has been a high-priority request from Member States. Investment cases are designed to help countries make their own economic rationales for action to prevent and control NCDs.

In April 2016, at the invitation of the Minister of Health, the United Nations Interagency Task Force on the Prevention and Control of NCDs led a joint United Nations team visit to Turkey (UNIATF, 2017). This aimed to promote a whole-of-government and whole-of-society response to NCDs and to support the government in putting NCDs at the forefront of the national development agenda.

 $^{^{\}scriptscriptstyle 1}$ "Presenteeism" is defined as reduced productivity at work.

² "Opportunity cost" is a term used in economics, defined as the cost of something in terms of an opportunity forgone: "opportunity cost is given by the benefits that could have been obtained by choosing the best alternative opportunity" (Oxford Dictionary of Economics [online]).

Fig. 1. Determinants of NCDs and responsibilities for response

Clinical management and secondary prevention Responsibility of Ministry of Health **Behavioural** Intermediate Main Underlying risk factors risk factors NCD determinants Unhealthy diet Overweight/obesity • Heart disease Poverty and poor living Physical inactivity Raised blood sugar Diabetes conditions Stroke Tobacco use High blood pressure Social exclusion Harmful alcohol Abnormal blood Cancer Design of cities and towns use lipids Chronic respiratory Availability and marketing disease of goods Prevention of NCD risk factors Responsibility of all ministries, Ministry of Health and society

In February 2017 the Minister of Health first raised the possibility of carrying out an NCD investment case study during a policy forum in Ankara, Turkey, on strengthening health financing to promote the country's health system goals (WHO Regional Office for Europe, 2017a). This was followed by a request from the Director-General of Public Health of the Ministry of Health to the WHO Regional Office for Europe for assistance in scaling up core NCD services in primary health care. This study was requested as evidence-based support for improvement of NCD interventions through employment of human resources, payments and incentives, as Turkey plans to strengthen the national model of chronic disease monitoring and to reduce the number of people registered at each family health centre.

A WHO visit to Turkey was undertaken in April 2017 to discuss the methodology and collect data. To support the overarching economic analysis, the team also carried out a desk review of published materials to develop a situation analysis and help identify policy strengths and areas for further development. Following the initial analysis, preliminary results were discussed at a national workshop in September 2017. It was agreed to expand the policy packages modelled to include policy interventions for all four risk factors and a package of clinical interventions, all of which were based on the WHO "best buys".

The investment case allows scaled-up action – and the costs of inaction – to be modelled in short-term (five years) and long-term (15 years) time frames. One scenario used is a continuation of the status quo, in which no new policies are implemented and current coverage levels remain in place – i.e. the costs of inaction. The other scenario is one in which selected policies and clinical interventions are scaled up over the next 15 years. The analysis used the WHO OneHealth Tool, an epidemiology-based population model developed by United Nations partners to enable strategic planning and costing of interventions and projection of the health benefits expected from their implementation. Health benefits are generated in terms of natural units (cases or deaths averted) but also monetized using the human capital approach to enable benefit–cost ratios (the primary ROI metric) to be evaluated and reported for each package of interventions.

Section 2 provides an analysis of NCD behavioural risk factors in Turkey, including current levels of tobacco, alcohol and salt consumption, as well as physical inactivity levels and prevalence of metabolic risk factors such as raised total cholesterol and raised blood pressure within the population. Section 3 outlines evidence-based policies and clinical interventions that can contribute to reducing the burden of disease – CVD in particular – and details the current implementation level of policies and interventions in Turkey. Section 4 describes the methods and tools used in the analysis. Section 5 presents the results, including total costs, and the expected health and economic benefits (such as healthy life-years gained, mortality averted and productivity gains) of implementing the policy packages described, as well as the clinical interventions. Section 6 outlines the conclusions to be drawn from these.

2. Situation analysis: NCDs and risk factors

This section sets out the background information used in preparation of the investment case analysis. It addresses NCDs as a whole and the extent to which behavioural risk factors – such as tobacco use, harmful alcohol consumption, physical inactivity and high salt intake – are present in Turkey, as well as the prevalence of metabolic risk factors such as raised blood pressure, cholesterol, obesity and diabetes.

The age-standardized premature mortality rate from the four major NCDs (cancer, CVD, diabetes and chronic respiratory disease) for those aged 30–69 years in Turkey in 2013 was 442 per 100 000 population for males and 211 per 100 000 for females; these rates are lower than the averages for the WHO European Region of 548 and 266, respectively (WHO Regional Office for Europe, 2017b). As of 2013, Turkish life expectancy at birth (81.71 years for women; 76.09 years for men) was above the average for the Region (80.86 years for women; 74.19 years for men).

Tobacco use

Findings from the 2012 Global Adult Tobacco Survey indicated that there were 14.8 million smokers in Turkey (Ministry of Health, 2014). Smoking prevalence declined between 2008 and 2012 for both men and women to become 41.5% for men and 13.1% for women by 2012. By 2017, however, 43.6% of men (aged 15 years and over) had become current smokers, with prevalence also rising among women (aged 15 years and over) to reach 19.7% (Ministry of Health, 2017a).

An estimated 10.4% of youths (12.1% of boys; 8.3% of girls) aged 13–15 years currently smoke cigarettes and 10.5% (13.6% of boys; 7.0% of girls) use other tobacco products (CDC, 2012). Almost half the students surveyed lived in homes where others smoked, and two thirds were around others who smoked in their presence outside the home.

According to a survey of chronic diseases and risk factors undertaken in 2011 (Ministry of Health, 2013), smoking was more prevalent in urban than rural areas, and was permitted in one in three houses (37%) and one in four workplaces (23%).

Key facts are summarized in Box 1.

Harmful alcohol use

Box 1. Tobacco snapshot

Tobacco use is of major concern as two fifths of Turkish men smoke – three times the proportion of women – and rates appear to have increased in the last five years, particularly for women.

Attributable NCDs include multiple forms of cancer; ischaemic heart disease (IHD), stroke and other CVD and circulatory diseases; chronic obstructive pulmonary disease and pneumoconiosis; and peptic ulcer disease, diabetes, cataract, macular degeneration and rheumatoid arthritis.

Already relatively low compared with the average in the WHO European Region, the rate of alcohol consumption appears to have reduced between 2004 and 2016 (WHO, 2018). Men are estimated to consume almost nine times (3.7 litres) as much pure alcohol per capita as women (0.4 litres) on average annually. In 2017, 13.1% of men (aged 15 years and over) and 3% of women (aged 15 years and over) are current alcohol drinkers – that

is, they had consumed alcohol in the previous 30 days (Ministry of Health, 2017a).

Harmful use of alcohol is higher among men than women. Among alcohol users, 8.7% of male drinkers binged (consumed six or more drinks in one sitting) during the month before the survey compared to 1.8% of female drinkers (Ministry of Health, 2017a).

Key facts are summarized in Box 2.

Physical inactivity

In 2017, the STEPwise approach to surveillance (STEPS) survey found that 33.1% of men and 53.9% of women were not meeting WHO recommendations on physical activity³ (Ministry of Health, 2017a). Rates of inactivity are higher for women than men; men do three times as many minutes of physical activity per day as women.

Around four fifths (81.3%) of adults do not engage in vigorous activity, particularly women.

Around a quarter (29%) of physical activity is work-related; over half (59.5%) is transport-related and an eighth is recreational (12.5%).

Key facts are summarized in Box 3.

Unhealthy diet (salt)

Salt/sodium consumption in Turkey is high. In 2008 salt intake was estimated to be 18 g a day (compared with the WHO recommendation to consume less than 5 g of salt per day), but this had reduced to 14.8 g per day in 2012 according to the SALTurk studies (Erdem et al., 2010; 2017). The STEPS survey (using a different method for estimation) found a level of 9.9 g per day in 2017 (Ministry of Health, 2017a).

The main source of salt is bread (34%). This is followed by salt added during cooking and preparing food before serving (30%), salt from various processed foods (21%) and salt added at the table during food consumption (11%) (Erdem et al., 2017). Around a quarter (26.0%) of the population add salt always or often when cooking or preparing food at home, and around a quarter (28.1%) add salt always or often before eating or when eating (Ministry of Health, 2017a).

Just under a fifth (17.1%) of cardiovascular deaths among 20–69-year-olds in 2010 were attributed to sodium consumption of more than the WHO recommendation to consume less than 2 g per day – a greater proportion for stroke than coronary heart disease (Mozaffarian et al., 2014).

Key facts are summarized in Box 4.

Box 2. Alcohol snapshot

Alcohol use is relatively low and appears to be reducing, but men drink almost nine times as much as women and have riskier drinking patterns.

Attributable NCDs include multiple forms of cancer, pancreatitis, epilepsy, diabetes, cirrhosis and IHD, stroke and other cardiovascular and circulatory diseases.

Box 3. Physical inactivity snapshot

Activity levels are of concern as one in three men and one in two women do not meet WHO recommendations on physical activity, and men do three times as much physical activity per day as women.

Attributable NCDs include coronary heart disease, type 2 diabetes and breast and colon cancers (Lee et al., 2012).

Box 4. Salt snapshot

Salt consumption is high, at around two to three times the WHO recommendation, but it is reducing. A quarter of people add salt to food before/ when eating it or when cooking.

Attributable NCDs include stomach cancer and increased risk of IHD, stroke and other cardiovascular and circulatory diseases due to hypertension.

The proportion of cardiovascular deaths attributable to high sodium is 17.1%.

³ 150 minutes of moderate-intensity physical activity per week, or equivalent

Metabolic risk factors

High levels of metabolic factors significantly increase the risk of having a cardiovascular event. Table 1 displays the prevalence of raised blood pressure,⁴ raised total cholesterol,⁵ raised blood sugar/diabetes⁶ and obesity within the Turkish population reported by the 2017 STEPS survey (Ministry of Health, 2017a). The prevalence of obesity (body mass index of 30 kg/m² or more) was found to be 28.8 (21.6 for men and 35.9 for women) in the population aged 15 years and over, for both sexes.

		Men			Women	
Factor	30–44 years	45–59 years	60-69 years	30–44 years	45–59 years	60-69 years
Raised blood pressure	16.3%	36.6%	54.8%	16.3%	41.8%	61.9%
Raised total cholesterol	21.8%	28.2%	34.1%	22.5%	50.9%	51.3%
Diabetes	7.8%	14.5%	30.6%	8.9%	18.4%	21.8%
Obesity	19.7%	32.5%	40.6%	30.1%	55.9%	66.9%

Table 1. Crude prevalence of metabolic risk factors, by age and gender

Source: Ministry of Health (2017a).

While elevated levels of any one factor can increase the risk of a cardiovascular event, the risk is compounded for individuals with multiple metabolic risk factors. WHO risk prediction charts assess the likelihood of an individual having a cardiovascular event and/or dying within 10 years by combining six factors: gender, age, blood pressure, cholesterol, smoking status and whether or not they have diabetes (WHO, 2016a). The prevalence of high cardiovascular risk among the Turkish population can be estimated from the 2017 STEPS survey according to the presence of risk factors or history of CVD or diabetes (Ministry of Health, 2017a). This suggests that 10.5% of Turkish adults aged 40–69 years have a probability of 30% or higher of having a fatal or nonfatal cardiovascular event within 10 years; this rises with age but differences between men and women are not statistically significant (Table 2).

Table 2. Crude prevalence of high cardiovascular risk, by age and gender

	Men		Women	
Factor	40–54	55–69	40–54	55–69
	years	years	years	years
10-year cardiovascular risk ≥30%, or with existing CVD	7.2%	22.5%	5.4%	11.6%

Source: Ministry of Health (2017a).

⁵ Raised total cholesterol ≥5.0 mmol/L or >=190 mg/dl or currently on medication for raised cholesterol.

⁴ Systolic blood pressure ≥140 mmHg and/or diastolic blood pressure ≥90 mmHg or currently on medication for raised blood pressure.

⁶ Raised blood glucose (defined as either plasma venous value of \geq 7.0 mmol/L (126 mg/dl) or capillary whole blood value of >=6.1 mmol/L (110 mg/dl)) or currently on medication for diabetes.

3. Policies and treatments to reduce the NCD burden

Endorsed in 2017, the Multisectoral action plan of Turkey on noncommunicable diseases 2017–2025 (Ministry of Health, 2017b) is a framework document encompassing the following current programmes:

- Republic of Turkey, Ministry of Health strategic plan 2013–2017;
- National cancer control programme 2013–2018;
- Prevention and control of chronic airway disease programme of Turkey 2014–2017;
- Prevention and control of cardiovascular disease programme of Turkey 2015–2020;
- Turkey diabetes programme 2015–2020;
- Prevention and control of kidney diseases programme of Turkey 2014–2017;
- Prevention and control of musculoskeletal system diseases programme of Turkey 2015–2020;
- National tobacco control programme action plan 2015–2018;
- Healthy nutrition and physical activity programme of Turkey 2014–2017;
- Programme for reducing high salt consumption in Turkey 2017–2021.

As highlighted in Section 1, WHO has published a menu of policy options and cost-effective interventions to prevent and treat NCDs (WHO, 2013; WHO, 2017b). The economic analysis for this investment case focuses on selected interventions for NCD prevention (policies on tobacco, alcohol, physical inactivity and salt) and management of cardiovascular risk and disease. The following sections summarize national efforts in implementing interventions for specific areas, drawing on the relevant published national and international documents.

Tobacco

Turkey ratified the WHO Framework Convention on Tobacco Control (FCTC) in 2004 and has committed to implementing a comprehensive tobacco control policy (WHO, 2017c; 2017d).

Table 3 summarizes a comparison of Turkey's current tobacco control measures against the MPOWER intervention package (monitor tobacco use and prevention policies; protect people from tobacco smoke; offer help to quit tobacco use; warn people about the dangers of tobacco; enforce bans on tobacco advertising, promotion and sponsorship; raise taxes on tobacco) as reported in the WHO report on the global tobacco epidemic (WHO, 2017e), supplemented by available national and international information.

Policy	Achievements (maximum of 4)	Current state of implementation
Monitor tobacco use and prevention policies	4	Recent and representative data are available for both adults and youths – for example, from the Global Youth Tobacco Survey (CDC, 2012), the Global Adult Tobacco Survey (Ministry of Health, 2014), the survey of chronic diseases and risk factors (Ministry of Health, 2013) and the STEPS survey (Ministry of Health, 2017a).
Protect people from tobacco smoke	4	For eight categories of public place ^a WHO reports that complete smoke-free laws exist (WHO, 2017d). Smoking violations incur fines for the smoker and the establishment and there are dedicated funds for enforcement.

Table 3. The current state of tobacco control measures in Turkey

Table 3 contd

Policy	Achievements (maximum of 4)	Current state of implementation
Offer to help to quit tobacco use	4	Nicotine replacement therapy is on the national essential medicines list and costs are fully covered. It is available without prescription in pharmacies. Tobacco cessation services are available in some health clinics and hospitals with costs fully covered. A toll-free telephone quit line exists.
Warn about the dangers of tobacco	4	Large health warnings on tobacco packages with all appropriate characteristics exist, covering 65% of the front and rear of the package combined. A national anti-tobacco campaign was aired during 2014–2016.
Enforce bans on tobacco advertising, promotion and sponsorship	4	WHO reports a ban on national television, radio and print media, as well as other forms of direct and/or indirect advertising. A point-of-sale ban on advertising is not in place.
Raise taxes on tobacco	4	WHO reports that total taxes comprised 82.4% of the retail price for most sold brands in 2016 and that cigarettes have become less affordable since 2008. Of the total taxes, 1.9% comprised specific excise tax, 65.3% was ad valorem excise tax and 15.3% was value added tax. According to WHO recommendations, the amount of total tax per pack should comprise at least 75% of the retail price including a 70% specific excise tax.

^a Legislation was assessed to determine whether smoke-free laws provided for a complete indoor smoke-free environment at all times, in all the facilities of each of the following eight categories of place: health care facilities; educational facilities other than universities; universities; government facilities; indoor offices and workplaces not considered in any other category; restaurants or facilities that serve mostly food; cafés, pubs and bars or facilities that serve mostly beverages; and public transport. *Source*: WHO (2017d).

Most of these policy interventions are also WHO "best buys" (WHO, 2017b); that is, effective interventions with cost—effectiveness analysis ≤100 international dollars per disability-adjusted life-year averted in low/middle-income countries. This list largely corresponds with those listed within the OneHealth Tool that can be modelled as part of the ROI analysis:

- monitor tobacco use/prevention policies
- protect people from tobacco smoke
- offer to help quit tobacco use: mCessation
- warn about danger: warning labels
- warn about danger: mass-media campaign
- enforce bans on tobacco advertising
- enforce youth access restriction
- raise taxes on tobacco
- plain packaging of tobacco products.

Alcohol

The global strategy and European action plan to reduce the harmful use of alcohol, as well as the updated Appendix 3 of WHO's global action plan for the prevention and control of NCDs 2013–2020, list core policy options for alcohol control (WHO, 2010; 2017b; WHO Regional Office for Europe, 2012). These are reproduced in Table 4, alongside some of the achievements in reducing alcohol consumption in Turkey. This assessment draws on various national and international sources.

Policy	Policy options	Current state of implementation
Taxation	Increase excise taxes on alcoholic beverages	Regular twice-yearly excise tax increases adjusted to the consumer price index were introduced on beer, wine and spirits in 2012. There are no taxes, however, to make alcohol products less attractive to young people. This area was scored as "limited" in one WHO report (WHO Regional Office for Europe, 2014) but as "fully achieved" by 2017 in another (WHO Regional Office for Europe, 2017c).
Advertising	Enact and enforce bans or comprehensive restrictions on exposure to alcohol advertising (across multiple types of media)	A full ban on alcohol marketing is well enforced. The law restricts the advertisement and promotion of alcohol. Alcohol manufacturers cannot advertise or sponsor organizations. This area was scored as "extensive" in one WHO report (WHO Regional Office for Europe, 2014), and as "fully achieved" in another (WHO Regional Office for Europe, 2017c).
Availability	Enact and enforce restrictions on the physical availability of retailed alcohol (via reduced hours of sale)	A decree restricting the sale and service of alcohol products entered into force in 2011. Retail sales are banned between 10pm and 6am. Alcohol cannot be displayed in shop windows and cannot be sold in the vicinity of schools and places of worship. All governmental and educational institutions are free of alcohol. Sale to children aged under 18 years is banned; this is effectively enforced. Any establishment that violates the regulation is punished severely, although it does not lose its licence. This area was scored as "moderate/extensive" in a WHO report (WHO Regional Office for Europe, 2014) and as only "partially achieved" in another (WHO Regional Office for Europe, 2017c).
Drink–driving	Enact and enforce drink– driving laws and blood alcohol concentration limits via sobriety checkpoints	In 1997, 0.50g/litre was set as the maximum blood alcohol content for drivers. With a new regulation introduced in 2013 the limit was decreased to 0.21 g/ litre for all drivers except for private car drivers. This area was scored as "limited/moderate" in a WHO report (WHO Regional Office for Europe, 2014).

Table 4. The current state of alcohol control interventions in Turkey

Table 4 contd

Policy	Policy options	Current state of implementation
Brief interventions	Provide brief psychosocial intervention for persons with hazardous and harmful alcohol use	Training for health care staff in screening and brief interventions for alcohol use is available.

The first three interventions listed in Table 4 are also WHO "best buys". The fourth and fifth are WHO "effective interventions" with cost–effectiveness analysis >100 international dollars per disability-adjusted life-year averted in low/middle-income countries. These policy interventions largely correspond with those listed within the OneHealth Tool that can be modelled as part of the ROI analysis:

- enforce restrictions on availability of retailed alcohol
- enforce restrictions on alcohol advertising
- enforce drink-driving laws (sobriety checkpoints)
- raise taxes on alcoholic beverages.

Physical inactivity

The updated Appendix 3 of WHO's global action plan for the prevention and control of NCDs 2013–2020 lists several policy options for improving physical activity levels (WHO, 2017b). These are reproduced in Table 5, alongside some of the achievements in increasing physical activity in Turkey.

Policy	Policy options	Current state of implementation	
Knowledge	Implementation of public awareness and motivational communications for physical activity, including mass-media campaigns for physical activity behaviour	Turkey has implemented at least one recent national public awareness programme on physical activity. Special days and weeks are celebrated in all 81 provinces to create awareness of nutrition, obesity, diabetes and physical activity. This intervention area was assessed as "moderate" in one WHO report (WHO Regional Office for Europe, 2014) and as "fully achieved" in another (WHO Regional Office for Europe, 2017c).	
Health system	Provision of physical activity counselling and referral as part of routine primary health care services through the use of a brief intervention	Initiatives for workforce development for nutrition and physical activity are under way, with nutrition and physical activity considered a priority element in primary care.	
Environment	Ensuring that macro-level urban design incorporates the core elements of residential density, connected street networks that include sidewalks, easy access to a diversity of destinations and access to public transport	National guidelines on physical activity, which include national recommendations, have been completed. The Turkish Association of Healthy Cities	
	Provision of convenient and safe access to high-quality public open space and adequate infrastructure to support walking and cycling	 (49 members covering 30 million people) is active in healthy urban planning and designing cities to promote active living (WHO Regional Office for Europe, 2015a). 	

Table 5 contd

Policy	Policy options	Current state of implementation	
Setting	Implementation of a whole-of-school programme that includes high-quality physical education, availability of adequate facilities and programmes to support physical activity for all children	Measures have been taken to promote physical activity in schools, including developing a physical education curriculum.	
	Implementation of multicomponent workplace physical activity programmes	-	
Promotion	Promotion of physical activity through organized sport groups and clubs, programmes and events	Physical activity is actively promoted in cities (WHO Regional Office for Europe, 2015a).	

The OneHealth Tool can model the following policy change as part of the ROI analysis:

• public awareness campaigning on physical activity.

Unhealthy diet

Policies on diet, physical activity and counteracting obesity were reviewed by WHO in 2016 (WHO Regional Office for Europe, 2016a). As the OneHealth Tool is not yet able to calculate the impact of interventions on fats and sugar, this section focuses on salt only.

Salt-reduction policies have been assessed overall as partially implemented (WHO Regional Office for Europe, 2017c). Table 6 compares Turkey's current state against SHAKE, a set of WHO measures that outline steps countries can take to reduce salt intake (surveillance; harness industry; adopt standards for labelling and marketing; knowledge; environment) (WHO, 2016b). Salt-reduction strategies in Turkey were assessed as "moderate/extensive" in a 2014 WHO report (WHO Regional Office for Europe, 2014) and as "fully achieved" in a 2017 WHO report (WHO Regional Office for Europe, 2017c).

Policy	Description ^a	Current state of implementation
Surveillance: measure and monitor salt use	Measure and monitor population salt consumption patterns and the sodium content of food	Turkey is one of the few countries that have conducted two nationally representative studies of 24-hour sodium urinary excretion (in 2008 and 2012). Further, the 2017 STEPS survey (Ministry of Health, 2017a) asked about salt consumption patterns and included urine spot tests for estimating salt consumption (see details in Section 2). Analysis of the nutritional composition of commonly available, widely consumed foods was carried out through an unpublished WHO survey. Better monitoring of the food supply has been recommended by WHO (WHO Regional Office for Europe, 2016a).

Table 6. The current state of	policies to reduce salt consum	ption in Turkey

Table 6 contd

Policy	Description ^a	Current state of implementation
Harness industry: promote reformulation of foods and meals to contain less salt	Set target levels for the amount of salt in foods and meals and implement strategies to promote reformulation	In 2012 a regulation to reduce the salt content of bread to 1.5 g/100 g was passed, and a 25% reduction of salt in bread was achieved. In addition, during 2012–2015 the regulation was extended beyond bread to limit the salt content of pastrami (reduction from 8.5 g to 7 g/100 g achieved), dried red pepper (22% reduction achieved), tomato paste (64% reduction achieved) and cheese (35–61% reduction achieved, depending on type) (WHO Regional Office for Europe, 2016a).
Adopt standards for labelling and marketing: implement standards for effective and accurate labelling and marketing of food	Adopt front-of-pack nutrition labelling systems (e.g. colour-coded for salt content level, "high salt" warning)	There is a compulsory message on salt packages: "Reduce salt, maintain your health".
Knowledge: educate and communicate to empower individuals to eat less salt	Implement integrated education and communication strategies to raise awareness about the health risks and dietary sources of salt in order to change behaviour	There are education and awareness-raising campaigns for the public about salt and health; these focus more on the discretionary addition of salt by consumers, notably in cooking at home, rather than the "hidden" salt content of processed foods.
Environment: support settings to promote healthy eating	Implement multicomponent salt- reduction strategies in community settings (e.g. schools, workplaces, hospitals)	Salt shakers have been removed from the cafeterias and food facilities of all public institutions.

^a Information in the Description column is derived from the SHAKE technical package for salt reduction (WHO, 2016b).

Four of these interventions are assessed as WHO "best buys" (reformulation; environment; knowledge; labelling). These policy interventions correspond with those listed within the OneHealth Tool that can be modelled as part of the ROI analysis:

- surveillance
- harness industry for reformulation
- adopt standards: front-of-pack labelling
- adopt standards: strategies to combat misleading marketing
- knowledge: education and communication
- environment: salt-reduction strategies in community-based eating spaces.

CVD and diabetes

The updated Appendix 3 of WHO's global action plan for the prevention and control of NCDs 2013–2020 lists multiple clinical interventions for CVD and diabetes (WHO, 2017b). A selection of those most relevant to this analysis is reproduced in Table 7, alongside an assessment of the situation in Turkey.

Policy	Description	Current state of implementation
	Screening for risk of CVD/ diabetes	Detection of diabetes, hypertension and cardiovascular risk stratification is done at the primary health care level. The national programme aims to measure arterial blood pressure for everyone aged 18 years and over to identify hypertensive individuals and those at risk of diabetes. A pilot project to implement cardiovascular risk assessment and diabetes screening has been piloted in 88 family health units and evaluated to help inform national scale-up.
Cardiovascular risk assessment and management	Provision of drug therapy (including glycaemic control for diabetes mellitus and control of hypertension using a total risk approach) and counselling to individuals who have had a heart attack or stroke and to people with high risk (≥30%) of a fatal or nonfatal cardiovascular event in the next 10 years	The national programme plans cardiovascular risk assessment for everyone aged 40 years and over, using the European Society of Cardiology's Systematic Coronary Risk Evaluation charts (ESC, 2012) and providing health recommendations according to identified risk. National guidelines for diabetes, hypertension and CVD risk assessment have been prepared, as well as simplified primary care protocols for hypertension and CVD risk assessment. Availability of cardiovascular risk assessment and management was assessed (based on self-reported data) as "not achieved" in 50% or more of primary health care facilities in a WHO survey in 2017 (WHO Regional Office for Europe, 2017c). Nevertheless, an estimated 50.7% of men and 46.8% of women at high cardiovascular risk were receiving drug therapy and counselling to prevent heart attacks and strokes according to the 2017 STEPS survey (Ministry of Health, 2017a), just short of the 50% global target. This also found that 72.9% of men and 77.5% of women with diagnosed hypertension were taking regular medication prescribed by a doctor.

Table 7. The current state	of clinical policies to	o reduce cardiovascul	ar risk in Turkev
	or entried ponetes t		

Table 7 contd

Policy	Description	Current state of implementation
	Treatment of new cases of AMI with either acetylsalicylic acid or acetylsalicylic acid and	Acute care of heart attacks and strokes appears to be relatively strong, following international practice and achieving international standards in places.
	clopidogrel, or thrombolysis or primary percutaneous coronary interventions	Turkey participated in the Stent for Life project and increased the frequency of primary percutaneous coronary interventions dramatically (Ural and Kayıkçıoğlu, 2014).
infarction str (AMI) and thr stroke Tre est	Treatment of acute ischaemic	According to 2012 data, there are 215 percutaneous coronary intervention-capable centres in Turkey, giving a crude number of centres per 1 million inhabitants of 2.86.
	stroke with intravenous thrombolytic therapy	The frequency of intravenous thrombolysis for stroke has been growing steadily in Turkey. Between 2006 and 2013 there was a fourfold increase in registered cases (Kutluk et al., 2016).
	Treatment of cases with established IHD and post myocardial infarction	Measures to prevent a further AMI or stroke (secondary prevention) exist in principle. Nevertheless, it has been suggested that cardiac rehabilitation is not achieving its full potential (Ural and Kayıkçıoğlu, 2014). The STEPS survey indicated that 5.6% of adults (aged 15 years and over) were taking aspirin to prevent or treat heart disease and 1.8% were taking statins.
	Glycaemic control	The national programme aims to measure blood glucose for everyone aged 40 years and over to identify diabetic individuals. Insulin and diabetic medication are available with full reimbursement.
Diabetes	Diabetic retinopathy screening and foot care to avoid complications	Foot care and eye checks are included as part of the standard follow-up of diabetic patients. A diabetes register monitors complications. Dilated fundus examination and foot vibration perception by tuning fork are basic technologies available in primary care (WHO, 2016c).

The OneHealth Tool is able to model the following package of interventions as part of the ROI analysis:

- screening for risk of CVD/diabetes
- treatment for those with high absolute risk of CVD/diabetes (>30%)
- treatment of new cases of AMI with aspirin
- treatment of cases with established IHD and post myocardial infarction
- treatment for those with established cerebrovascular disease and post stroke
- treatment of cases with rheumatic heart disease (with benzathine penicillin)
- standard glycaemic control
- intensive glycaemic control
- retinopathy screening and photocoagulation
- neuropathy screening and preventive foot care.

Summary

This review of current NCD interventions at the policy and individual service levels uncovered gaps in implementation of the WHO-recommended cost-effective NCD preventive and clinical interventions and drew attention to areas that need strengthening and scale-up to achieve 100% coverage. It also highlighted ways to increase the level of intensity if coverage was already 100%. The estimation of current levels of coverage based on the assessment above is summarized in Table 8. This shows the current status of coverage; the current level of intensity is assumed to be level 2. In the next section, the OneHealth Tool models 100% coverage for all interventions and the highest level (4) of intensity.

Торассо	
Monitor tobacco use/prevention policies	100%
Protect people from tobacco smoke	100%
Offer to help quit tobacco use: mCessation	100%
Warn about danger: warning labels	100%
Warn about danger: mass-media campaign	100%
Enforce bans on tobacco advertising	100%
Enforce youth access restriction	100%
Raise taxes on tobacco	100%
Plain packaging of tobacco products	N/A
Harmful alcohol use	
Enforce restrictions on availability of retailed alcohol	100%
Enforce restrictions on alcohol advertising	100%
Enforce drink-driving laws (sobriety checkpoints)	100%
Raise taxes on alcoholic beverages	100%
Physical activity	
Public awareness campaigning on physical activity	100%
Salt	
Surveillance	100%
Harness industry for reformulation	70%
Adopt standards: front-of-pack labelling	40%
Adopt standards: strategies to combat misleading marketing	50%
Knowledge: education and communication	50%
Environment: salt-reduction strategies in community-based eating spaces	40%
Clinical interventions: CVD	
Screening for risk of CVD/diabetes	25%
Treatment for those with high absolute risk of CVD/diabetes (>30%)	25%
Treatment of new cases of AMI with aspirin*	90%
Treatment of cases with established IHD and post myocardial infarction*	90%

Table 8. Estimation of current level of coverage of NCD interventions to be costed within the OneHealth Tool

Table 8 contd

Treatment for those with established cerebrovascular disease and post stroke*	
Clinical interventions: diabetes	
Standard glycaemic control	47%
Intensive glycaemic control	28%
Retinopathy screening and photocoagulation	0.1%
Neuropathy screening and preventive foot care	44.1%

Notes: Coverage of policy interventions was estimated by the national team of experts at the Directorate-General of Public Health except for those interventions marked with an asterisk, which were estimated by the authors. Although coverage for the intervention "plain packaging of tobacco products" was not estimated separately, the OneHealth Tool calculates the whole effect of a tobacco control policy package that includes plain packaging.

4. Methods

To make informed decisions about which policies and clinical interventions to implement to reduce the burden of NCDs, it is essential that policy-makers understand the expected benefits from and costs of investing in NCD interventions in their countries. Against this background, a WHO team visited Turkey in April 2017 to conduct an investment case analysis, with a focus on salt reduction and clinical intervention policy packages. A national team was appointed to assist WHO in data collation and analysis. The preliminary analysis was discussed at a meeting of epidemiologists, statisticians, health economists and other staff of the Directorate-General of Public Health and Ministry of Health on 29 September 2017. The decision was taken to expand the work to the full scope of policy and clinical packages. The full analysis was completed in December 2017.

This section outlines the different methods and economic models applied at different stages in the economic analysis:

- calculation of economic burden of NCDs in terms of direct costs and indirect costs (absenteeism, presenteeism and premature death);
- costing of interventions (clinical and policy interventions);
- assessment of health impacts; and
- ROI analysis.

At the request of the national team, an additional calculation included a health care resources utilization analysis of the ambulatory care-sensitive conditions (ACSCs) among the four main NCDs covered by social security funds, hospitalizations and primary care visits.

Calculation of economic burden of NCDs

The NCD economic burden model applied was developed by WHO and the United Nations Development Programme, and provides estimates of the current direct and indirect costs of NCDs in Turkey. Data sources for calculation of the economic burden are listed in Annex 2. The data used on population by age and sex for the period 2016–2031 were based on the intermediate population growth scenario of the Turkish Statistical Institute (2013) (Table A3 in Annex 3) and the cohort method with most recent data. Details incorporated were incidence rates by age and sex for heart attack and stroke; and prevalence rates by age and sex for diabetes, hypertension and chronic respiratory disease. Mortality rates by age and sex were applied for each condition. The model calculated projections for incidence, prevalence and mortality for diabetes, CVD and chronic respiratory disease between 2016 and 2031, holding current rates constant.⁷ These projections were summarized as total incidence, prevalence and mortality for both the entire population and the working-age population, defined as those aged 15–64 years.

The following steps were carried out to calculate the economic costs.

- As only total government health expenditure data are available in Turkey, the share of total health expenditure on NCDs was calculated on the basis of a WHO analysis covering 13 countries (Garg & Evans, 2011; see Table A4.1 in Annex 4).
- The annual value (in terms of economic output) of each full-time worker in Turkey was calculated. This is based on gross domestic product (GDP) per employed person, defined as the country's GDP divided by its total employed labour force. To arrive at the total employed labour force for Turkey, National Statistical Office data on the total labour force aged 15 years and over, the unemployment rate and the labour force participation rate were used.
- Data were incorporated on the extent to which NCDs reduce worker productivity. From the academic literature (Anesetti-Rothermel & Sambamoorthi, 2011; Wang et al., 2003), rates were found to describe (a) the reduction in labour force participation due to hypertension, stroke, AMI and diabetes; (b) the reduction in full-time hours worked due to absenteeism; (c) the reduction in productivity due to presenteeism; and (d) the total time taken to replace a worker (see Table A4.2 in Annex 4).
- The exact number of people with NCDs working in Turkey in 2016 was determined. Using the labour force participation, unemployment and mortality rates, the model began with Turkish people of working age with NCDs; subtracting those who chose not to participate in the labour force or were unemployed; subtracting those who could not participate in the labour force specifically because of their NCD; and finally, subtracting those who died. The result was an estimate of active workers with NCDs.
- The final steps were to calculate the costs of absenteeism and presenteeism for surviving active workers with NCDs. The model applied the relevant productivity figures found in the second step to the populations determined in the third step and multiplied this by GDP per employed person. This calculation resulted in the total indirect costs of each NCD.

Calculation of policy and interventions costs

- Costs of policy interventions were calculated using the WHO Costing Tool (WHO, 2012). This identifies, quantifies and values each resource required for the intervention as follows:
 - for each policy, the Tool costs human resources, training, external meetings, mass-media campaigns (e.g. television and radio time, newspaper ads) and other miscellaneous equipment needed to enact policies and programmes;
 - each policy contains assumptions, set by WHO experts, about the quantity of inputs required to implement and enforce it – the Tool estimates the quantity of resources needed at the national, regional and district levels;
 - unit costs for resource items are taken from the WHO-CHOICE database (WHO, 2017f).
- The costs of clinical interventions were calculated using the OneHealth Tool, which conveniently has built-in functionality that works out expected costs of treatment interventions. For each intervention, the OneHealth Tool takes as input data points such as the salaries of medical staff and the quantities of drugs and supplies needed, as well as their prices.

⁷ The model estimates growth in prevalence, incidence and mortality due to population growth only – not growth in disease rates.

ROI

ROI is a performance measure used to evaluate the efficiency of health care investment. It compares the magnitude and timing of benefits from health intervention directly with the magnitude and timing of investment costs. ROI is the ratio of the discounted (present) value of the benefits to the investment costs. Future benefits are discounted since a unit of currency in the future is worth less than a unit today owing to time value of money.

An ROI analysis, based on an Excel model developed by WHO for this analysis, provided estimates for economic gains that accrue from investing in the set of cost-effective interventions identified during the visit. The policy-based interventions included in this calculation are listed in Table 7 above.

The methodology used is the NCD ROI model developed in 2015 for use by the United Nations Development Programme/WHO Joint Programme on Governance of NCDs using the OneHealth Tool and WHO Costing Tool. More detail on use of the tool is available from the OneHealth Tool Manual (Avenir Health, 2017) and it is discussed in detail in the technical appendix to the forthcoming RTI International report, *The investment case for noncommunicable disease prevention and control* (RTI International, in press).

To determine the overall impact of the set of interventions, productivity measures were assessed using the following steps.

- Data on the amount by which NCDs reduce worker productivity were incorporated, as noted for the NCD economic burden model. As interventions reduce the projected incidence of IHD and stroke, there is an associated increase in the number of healthy life-years of the population. By considering the increase in healthy life-years, GDP per employed person and the reduction in rates for absenteeism and presenteeism, an increase in GDP can be determined, attributed to the value of avoided absenteeism and presenteeism.
- By considering the labour force participation rate in Turkey and the projected number of Turkish deaths avoided, the increase in labour force participation due to avoided deaths was calculated. An increase in economic output was therefore attributed to the value of avoided mortality.
- The final economic gain came from the reduced time spent having to seek new workers for replacement. The academic literature estimates the time taken to replace workers to be around 10 weeks, on average. The worker replacement rate, applied to both the total deaths avoided and the increase in healthy lifeyears due to avoided IHD and stroke, gives the increase in GDP resulting from not having to replace staff so frequently.

ROI rates were calculated for the interventions listed above in Table 7. These were selected on the basis of data availability to ensure sufficient data for calculations of costs and health impacts.

The projected economic gains from implementing the cost-effective interventions were therefore the value of avoided presenteeism, the value of avoided absenteeism and the value of avoided mortality. The impact of an intervention, measured as the total increase in GDP, was calculated by combining the three types of gain.

The ROI for Turkey was arrived at by comparing the impact (increase in GDP) of the interventions with the total costs of setting up and implementing the interventions. It was calculated using the net present value approach to future costs and economic gains, with 3% discounting.

Resource utilization for ACSCs

The aim of this analysis was to identify the elements of primary care that need strengthening to avoid unnecessary hospitalization of ACSCs. ACSCs are health conditions for which hospitalization or emergency care can be avoided by addressing them effectively in primary care. WHO developed a methodology of analysis; this

has already been applied in several countries, including Germany, Kazakhstan, Latvia, Portugal and the Republic of Moldova (WHO Regional Office for Europe, 2015b; 2015c; 2015d; 2015e; 2016b). The analysis here used only one component of the methodology, in a similar approach to that taken in Kazakhstan, to calculate the costs of potentially avoidable hospitalizations.

Data for this analysis include the number of hospitalizations, number of outpatient visits and costs of both types of care covered by the social security fund. The analysis did not included hospitalizations in private clinics, nor did the social security fund cover all costs of hospitalizations. The data were obtained from a joint study by the Public Hospitals Authority and the Social Security Institute (SSI) of Turkey. The data provided by the Ministry of Health indicate that about 30% of all patients in 2015 were served by university and private hospitals. Thus, the total health care payments by the SSI are multiplied by 1.3 in order to obtain total hospitalization costs.

The analysis included diseases considered to be ACSCs that are associated with the NCDs of interest; these are listed in Table 9, along with their International Classification of Disease, 10th revision (ICD-10) codes.

Table 9.	Diseases	used	within	the	ACSC	analysis
----------	----------	------	--------	-----	------	----------

Disease	ICD-10 codes
Hypertension	110.0–113.0
Angina	120.0–120.9
Heart Failure (congestive)	150.0–150.9
Diabetes	E11
Bronchial asthma	J45.0–J45.9
Chronic bronchitis and unspecified emphysema	J40.0–J43.9
Chronic obstructive pulmonary disease (COPD)	J44.0–J44.9

5. Results

This section provides an assessment of the economic burden of NCDs before summarizing the components of the ROI analysis – including health benefits, economic benefits and total costs – and discussing the ROI for each package of interventions.

Economic burden assessment

Direct costs

The estimate of direct costs of the economic burden considered only government health care expenditure, not non-health care costs such as transportation.

Total government health expenditure for Turkey in 2016 was TL 82.1 billion. As noted above, government health care spending on NCDs in Turkey was estimated based on National Health Account data on NCD spending in 13 other countries (Garg & Evans, 2011). Assuming consistency with these countries (all have a similarly high NCD disease burden, although some are high-income countries), 30% of government expenditure on health would be attributable to NCDs (13.4% on CVD; 7% on cancer; 6% on chronic respiratory disease; 4% on diabetes). Using international rather than country-specific numbers is a limitation of this study, considering the wide variability of the share of health care expenditure per disease group. Total health care expenditure on the four main NCDs is estimated to be TL 24.6 billion for 2016 (Fig. 2).





Indirect costs

For Turkey, indirect economic losses due to NCDs were modelled from reduced labour force participation, increased absenteeism and presenteeism and losses due to premature death.

The calculation of absenteeism and presenteeism is based on the surviving workforce. Results are shown in Fig. 3 and details of underlying numbers are given in Table A4.2 in Annex 4. Indirect costs could only be calculated for CVD and diabetes. The costs of absenteeism were TL 1.2 billion for CVD and TL 0.4 billion for diabetes. The total burden of presenteeism reached TL 2.8 billion.

Indirect costs of premature death in Turkey were estimated using the human capital method. This assumes that forgone economic output is equivalent to the total output that would have been generated by workers through the course of their life until reaching retirement age. The net present value approach was used. The cost of premature death was calculated by multiplying GDP per worker by the labour force participation rate, by the age-specific employment rate. Total cost of premature death was estimated at TL 21.7 billion (Fig. 4).

Cancer is the costliest of the four NCDs in terms of economic losses resulting from premature mortality. Diabetes does not appear to be a leading cause of premature death, despite the productivity losses in presenteeism; nevertheless, many people with diabetes may die rematurely of a cardiovascular event.

Total economic costs

Table 10 summarizes the total direct and indirect costs of NCDs in Turkey. Economic losses due to indirect costs are almost four times larger than those due to direct costs. The government's estimated spending on the four main NCDs is already TL 24.6 billion, but additional losses to the economy (absenteeism, presenteeism, premature death) amount to TL 45.1 billion. This would be even larger had it been possible to estimate the costs of absenteeism and presenteeism for cancer and respiratory diseases.



Fig. 3. Costs of absenteeism and presenteeism for CVD and diabetes, 2016

Fig. 4. Costs of premature death, 2016



Table 10. Economic burden of NCDs in Turkey (billion TL), 2016

CVD	Cancer	Diabetes	Chronic respiratory diseases	Total for all NCDs
11.0	5.5	3.3	4.8	24.6
1.2	N/A	0.4	N/A	1.6
7.9	N/A	13.9	N/A	21.8
6.5	14.1	0.2	0.9	21.7
15.5	14.8	14.5	0.94	45.1
26.5	19.6	17.8	5.7	69.7
	11.0 1.2 7.9 6.5 15.5	11.0 5.5 1.2 N/A 7.9 N/A 6.5 14.1 15.5 14.8	11.0 5.5 3.3 1.2 N/A 0.4 7.9 N/A 13.9 6.5 14.1 0.2 15.5 14.8 14.5	CVD Cancer Diabetes diseases 11.0 5.5 3.3 4.8 1.2 N/A 0.4 N/A 7.9 N/A 13.9 N/A 6.5 14.1 0.2 0.9 15.5 14.8 14.5 0.94

The total drag on the economy of Turkey is TL 69.7 billion, which is equivalent to 3.6% of GDP annually.

Fig. 5 shows the structure of the NCD economic burden in Turkey in 2016. Government health care expenditure represents 35.3% of all NCD-related costs, but these are just the tip of the iceberg for the NCD economic burden.

Fig. 5. Structure of the NCD economic burden in Turkey, 2016



Intervention cost assessment

Incremental intervention costs were estimated for the period 2018–2032. Table 11 shows costs for each of the first five years of this period, plus the five-year and 15-year totals for packages of interventions included in the analysis.

The CVD and diabetes clinical interventions produced the largest cost estimates. Treating those at high absolute risk of CVD and with clinical forms of CVD and diabetes costs TL 222.6 million in the baseline year and increases

to TL 443.3 million in 2022. Implementing the entire CVD and diabetes clinical intervention package over the five-year scale-up period would cost TL 1619 million; over 15 years it would cost TL 15 794 million.

Intervention package	2018	2019	2020	2021	2022	Total 5 years	Total 15 years
Tobacco control package	69.1	70.5	78.1	81.7	84.1	383.5	1 443.4
Alcohol control package	19.6	17.7	17.7	17.9	17.7	90.6	252.1
Physical activity awareness package	0.4	12.4	11.1	11.1	11.1	46.1	165.6
Salt-reduction package	13.6	26.1	26.8	28.2	29.6	124.3	527.6
All policy interventions, total	102.7	126.7	133.7	138.9	142.5	644.5	2 388.7
CVD and diabetes clinical intervention package	222.6	246.6	317.1	389.6	443.3	1 619	15 794.0
All interventions (policy + clinical), total	325.3	373.3	450.8	528.5	585.8	2 263.7	18 183.1

Table 11. Cost overview of packages of policy and clinical interventions (million TL), 2018–2022

The total costs per year for the tobacco, physical activity and salt packages were twice as cheap as those for the clinical package in the initial five years. The most costly policy package is tobacco: its total cost based on MPOWER guidelines is TL 69.1 million for the first year (2018).

Health benefit assessment

All interventions provide significant reductions in the number of lives lost to CVD-related causes. CVD clinical interventions have the greatest impact in terms of mortality cases (509 000 deaths averted), but the number of healthy life-years gained and numbers of strokes and acute IHD cases averted was highest in the salt-reduction package (Table 12). Alcohol control interventions had the smallest effect.

Intervention package	Strokes averted	IHD events averted	Mortality averted	Healthy life- years gained
CVD and diabetes clinical intervention package	99 871	34 751	509 624	1 136 560
Tobacco control package	59 443	70 607	66 375	567 316
Alcohol control package	1 856	388	1 332	10 233
Physical activity awareness package	1 935	5 805	3 382	29 224
Salt-reduction package	481 971	394 646	437 572	3 582 908

Table 12. Estimated health benefits over a 15-	vear time horizon

Each set of interventions also restores healthy life-years to the population. The CVD clinical interventions and tobacco control and salt reduction packages prevent strokes and cardiovascular events, and thus individuals avoid disabling states (such as partial paralysis from stroke) that can increase pain and suffering, reduce mobility and impair speech and thought.

Economic benefit assessment

NCDs included in this analysis are associated with a reduction in labour workforce and productivity due to premature mortality, fewer days of work (absenteeism) and reduced productivity while at work (presenteeism).

Fig. 6 demonstrates the labour productivity gains that result from the prevented deaths and disease cases over a 15-year period, as described in Table 12.





The biggest positive impact on productivity is from decreased mortality (84.7% of total productivity gains), followed by reduced presenteeism (8.1%) and absenteeism (7.2%). Policy packages and interventions in CVD and diabetes in primary care result in net present value TL 159.8 billion in labour productivity gains over 15 years (equivalent to 8.2% of Turkey's 2015 GDP).⁸

ROI assessment

Comparing the costs and benefits of packages of interventions shows that most of the NCD prevention interventions included in the analysis – for tobacco control, salt reduction and increasing physical activity – have positive ROIs in both the five-year and the 15-year periods (Table 13). The alcohol-control package did not demonstrate a positive ROI overall.

Salt interventions have the highest ROI: for every TL invested in the package of salt interventions, the expected return is TL 88 over a 15-year period (considering that interventions in this package need to increase in both coverage and intensity, while other policy interventions already have 100% coverage and mainly need to increase intensity). The tobacco package has the next highest ROI for 15 years (5), followed by the CVD and diabetes clinical intervention package (4.3). In the five-year period CVD and diabetes clinical interventions have the second highest ROI (3.3), followed by the tobacco package (2.6).

⁸ 2015 GDP was TL 1.95 trillion.

		5 years			15 years	
Intervention package	Total costs	Total productivity benefits	ROI	Total costs	Total productivity benefits	ROI
Tobacco control package	383.4	961.1	2.6	1 443.4	14 351.6	5.0
Alcohol control package	90.6	20.1	0.2	252.1	282.6	0.6
Physical activity awareness package	46.1	59.0	1.1	156.6	732.8	2.3
Salt-reduction package	124.3	6 815.9	51.0	527.2	93 873.7	88.0
CVD and diabetes clinical intervention package	1 619.0	4 931.8	3.3	7 877.0	47 527.6	4.3

Table 13. Costs, benefits and ROI at 5 and 15 years, by intervention package (million TL)

The package of alcohol interventions effectively provides no ROI for either the five-year or 15-year assessment periods. This may be because any interventions would need to cover the whole population, even though the problem of harmful alcohol consumption in Turkey affects a relatively small number of people.

Salt is the clear "best buy", offering the highest ROI over both five- and 15-year periods.

Hospitalizations and ambulatory care analysis

The cost analyses of hospitalizations (Table 14) and ambulatory care (Table 15) include only SSI data. The costs paid from other sources (out-of-pocket and so on) are not included in the analysis since no data are available on these in Turkey.

Table 14 presents the hospitalization costs and rates for the different NCDs identified to be included in the analysis. These include hypertension, angina, heart failure (congestive), diabetes, bronchial asthma, chronic bronchitis and unspecified emphysema, and COPD. The number of SSI-funded hospitalizations for these NCDs is 942 047. For each NCD, the proportion of registered cases hospitalized with SSI funding is generally not very high (range 0.3–3.3%).

According to different estimates, 60–80% of hospitalizations due to ACSCs are potentially preventable; thus, of 942 047 hospitalizations, an estimated 565 000–750 000 could be prevented. As the cost of SSI-funded hospitalizations for NCDs totals TL 420 649 702, preventing ACSCs when possible would deliver substantial savings.

Table 14. Hospitalization	n rates and costs	(2016)
---------------------------	-------------------	--------

Disease	ICD-10 codes	Total number of cases (registered)	Total number of hospitalizations paid by SSI	Proportion of hospitalizations (%)	Cost of hospitalizations (TL)
Hypertension	110.0-113.0	10 413 033	345 355	3.3	78 913 380
Angina	120.0-120.9	635 832	12 683	2.0	16 509 975
Heart failure (congestive)	150.0–150.9	1 308 476	32 706	2.5	36 382 225
Diabetes	E11-E11.9	4 085 461	130 624	3.2	45 076 864
Chronic bronchitis and unspecified emphysema	J40–J43.9	1 325 959	22 412	1.7	7 773 592
COPD	J44.0–J44.9	3 399 544	209 675	6.2	164 912 492
Bronchial asthma	J45.0–J45.9	7 251 746	188 604	2.6	71 081 174
Total		28 420 051	942 059		420 649 702

Table 15 gives the breakdown of costs for ambulatory care. The rate of ambulatory care for some NCDs is relatively low – for example, 19.1% of registered patients had ambulatory visits during 2016 for diabetes, including 4.5% visited family medicine centres (FMCs) (considering that the data show the total the number of visits and some patients had multiple visits, the actual proportion of patient could be even lower). For bronchial asthma 28.8% of registered patients visited primary care physicians (just 7.5% had ambulatory visits to FMCs); for angina only 8% visited primary care physician and for congestive heart failure it was 12.7%. A substantial share (79%) of hypertensive patients had primary care visits, however, so those with concurrent conditions (co-morbidities) – such as diabetes, angina and so on– might be in this group, since the primary reason for the visit would be coded as hypertension. Once again, the data show only those patients whose visits were covered by the SSI; the costs for those visiting private primary care physicians are not included. Nevertheless, the data seem to show underuse of primary care by patient with some ACSCs.

Table 15 shows the total costs of ambulatory care in both in hospitals and FMCs. This is estimate at TL 2.1 billion, of which only one third (0.7 trillion, 34%) is in the FMCs. From the whole costs of ambulatory care 51.9% is due to hypertension, 20.8% to bronchial asthma 17.4% to COPD, 1.1% to chronic bronchitis and unspecified emphysema, 7.1% to diabetes, 1.4% to heart failure and remaining 0.3% to angina.

Table 15. Ambulatory care rates and costs (2016)	latory care rate	s and costs (20	16)							
			Ambu	Ambulatory care in hospitals	hospitals	Amb	Ambulatory care in FMCs	FMCs		
		Total number	Number	Proportion of ambulatory visits to		Number of vicito 40	Proportion of ambulatory	4 	Total proportion of	
Disease	icD-IU codes	ot cases (registered)	or visits to hospitals	nospitals (%)	COSTS OT VISITS (TL)	ot visits to FMCs	visits to FMCs (%)	Losts of visits (TL)	ambulatory visits (%)	Iotal costs (TL)
Hypertension	110.0-113.0	10 413 033	3 730 862	35.8	565 399 406	4 499 721	43.2	505 465 369	79.0	1 070 864 775
Angina	120.0-120.9	635 832	48 343	7.6	6 406 635	2 328	0.4	265 961	8.0	6 672 596
Heart failure (congestive)	150.0-150.9	1 308 476	129 200	9.9	24 753 797	36 899	2.8	3 180 076	12.7	27 933 873
Diabetes	E11-E11.9	4 085 461	593 152	14.5	123 033 838	185 209	4.5	24 036 093	19.1	147 069 932
Chronic bronchitis and unspecified emphysema	J40–J43.9	1 325 959	156 309	11.8	17 328 155	90 788	6.8	5 864 245	18.6	23 192 401
СОРД	J44.0–J44.9	3 399 544	1 003 684	29.5	283 556 885	295 159	8.7	74 693 206	38.2	358 250 091
Bronchial asthma	J45.0–J45.9	7 251 746	1 545 289	21.3	334 545 951	544 187	7.5	94 365 282	28.8	428 911 233
Total		28 420 051	7 206 839		1 355 024 668	5 654 291		707 870 233		2 062 894 901

6. Conclusion

NCDs pose a significant threat to Turkish health and economic development. This report sets out the case for further investment in action against NCDs. It assesses the economic burden of NCDs for the country, costs specific interventions and presents a cost–benefit analysis for five intervention packages to demonstrate cost-effective solutions.

Analysis of the economic burden of NCDs in 2016 estimates that total economic losses to the economy are TL 69.7 billion annually, which is equivalent to 3.6% of GDP. Of the total costs 35.3% are from direct health expenditure but the majority of the economic burden results from indirect costs, predominantly due to presenteeism and premature mortality.

Five packages of interventions were economically evaluated – four to reduce the prevalence of NCD behavioural risk factors and one for clinical interventions. For the intervention packages on tobacco, alcohol and physical inactivity, the impact and costs of increasing intensity levels were estimated. For the salt reduction package, for which coverage is not yet 100%, the costs of increasing both coverage and intensity were estimated. Increasing the intensity of policy packages to reduce the consumption of tobacco, alcohol and to increase physical activity was estimated to cost TL 383.5 million, TL 90.6 million, TL 46.1 million at five years, respectively. The cost of increasing the intensity and coverage of the salt-reduction package at 5 years would be TL 124.3 million. The CVD and diabetes clinical interventions were found to be the most expensive options, costing TL 1619.0 million at five years.

The ROI analysis found an ROI greater than one at both five and 15 years (i.e. a good investment from an economic perspective) for every intervention package assessed, except that for alcohol control. By far the most cost-effective interventions in Turkey are those for salt. The economic benefits of this package far exceed implementation costs in the short (five years) and long term (15 years). The salt reduction policy package achieved an ROI of 51 and 88 TL at five and 15 years, respectively. Nevertheless, the WHO "best buys" interventions for tobacco, physical activity and clinical interventions remain cost-effective, despite lower ROIs in Turkey relative to salt. Only the alcohol control package did not achieve an ROI greater than one at five or 15 years: this may be because the problem of harmful alcohol consumption in Turkey is relatively low scale.

Analysis of the ACSCs revealed that 942 047 hospitalizations occur annually for such conditions among the four main NCDs of interest (paid by SSI; the number of hospitalizations for private clinics is unknown). The rate of hospitalizations relative to the number of registered patients is comparable with other countries where such analysis has been performed. Preventing ACSC hospitalizations when possible would deliver substantial savings.

In conclusion, there is much to be gained in terms of health and economic benefits from a focus on NCD prevention. This would require maintenance of the already high levels of implementation of policy packages such as that for tobacco control, with increased attention to intensity – for example, with enforcement of existing regulations – as well as further investment in areas of relatively low current coverage, such as salt reduction. Such actions are cost-effective and relatively cheap compared to the indirect and direct costs NCDs cause under status quo conditions. While implementation of the intervention packages will require engagement from sectors beyond health – such as finance, economy and agriculture – the benefits from the investments would accrue across the whole of government and of society.

References

Anesetti-Rothermel A, Sambamoorthi U (2011). Physical and mental illness burden: disability days among working adults. Popul Health Manag. 14(5):223–30.

Avenir Health (2017). OneHealth Tool [website]. Glastonbury, CT: Avenir Health (http://www.avenirhealth.org/ software-onehealth.php, accessed 25 July 2017).

Bloom DE, Cafiero ET, Jané-Llopis E, Abrahams-Gessel S, Bloom LR, Fathima S et al. (2011). The global economic burden of non-communicable diseases, Geneva: World Economic Forum (http://apps.who.int/medicinedocs/en/d/Js18806en/, accessed 21 August 2017).

CDC (2012). Global youth tobacco survey factsheet: Turkey. Atlanta, GA: Centers for Disease Control and Prevention (https://nccd.cdc.gov/GTSSDataSurveyResources/Ancillary/DataReports.aspx?CAID=1, accessed 28 September 2017).

Erdem Y, Arici M, Altun B, Turgan C, Sindel S, Erbay B et al. (2010). The relationship between hypertension and salt intake in Turkish population: SALTURK study. Blood Press. 19(5):313–8. doi:10.3109/08037051003802541.

Erdem Y, Akpolat T, Derici Ü, Şengül Ş, Ertürk Ş, Ulusoy Ş et al. (2017). Dietary sources of high sodium intake in Turkey: SALTURK II. Nutrients. 9(9):E933 (http://www.mdpi.com/2072-6643/9/9/933, accessed 28 September 2017).

ESC (2012). SCORE risk charts: the European cardiovascular disease risk assessment model [website]. Brussels: European Society of Cardiology (https://www.escardio.org/Education/Practice-Tools/CVD-prevention-toolbox/ SCORE-Risk-Charts, accessed 6 April 2017).

Garg CC, Evans DB (2011). What is the impact of noncommunicable diseases on national health expenditures: a synthesis of available data. Geneva: World Health Organization (NCD discussion paper no. 3; http://www.who. int/healthsystems/NCDdiscussionpaper3.pdf, accessed 23 August 2017).

Kutluk K, Kaya D, Afsar N, Arsava EM, Ozturk V, Uzuner N et al. (2016). Analyses of the Turkish National Intravenous Thrombolysis Registry. J Stroke Cerebrovasc Dis. 25(5):1041–7. doi:10.1016/j. jstrokecerebrovasdis.2016.01.015.

Lee IM, Shiroma EJ, Lobelo F, Puska P, Blair SN, Katzmarzyk PT (2012). Effect of physical inactivity on major non-communicable diseases worldwide: an analysis of burden of disease and life expectancy. Lancet. 380(9838):219–29.

Ministry of Health (2013). Chronic diseases and risk factors survey in Turkey. Ankara: Ministry of Health (http://sbu.saglik.gov.tr/Ekutuphane/Yayin/463, accessed 27 September 2017).

Ministry of Health (2014). Global Adult Tobacco Survey: Turkey 2012. Ankara: Ministry of Health (http://www. who.int/tobacco/surveillance/survey/gats/tur/en/, accessed 28 September 2017).

Ministry of Health (2017a). National household health survey in Turkey: prevalence of noncommunicable disease risk factors, 2017. Final report. Ankara: PGlobal Global Advisory and Training Services.

Ministry of Health (2017b). Multisectoral action plan of Turkey on noncommunicable diseases 2017–2025. Ankara: Ministry of Health (http://www.euro.who.int/en/countries/turkey/publications/multisectoral-action-plan-of-turkey-for-noncommunicable-diseases-2017-2025-english-and-turkish, accessed 19 December 2017).

Mozaffarian D, Fahimi S, Singh GM, Micha R, Khatibzadeh S, Engell RE et al. Global sodium consumption and death from cardiovascular causes. N Engl J Med. 2014;371(7):624–34.

RTI International (in press). The investment case for noncommunicable disease prevention and control: technical appendix. Geneva: World Health Organization.

Turkish Statistical Institute (2013). Causes of mortality statistics, ages 30–70 [database]. Ankara: Turkish Statistical Institute (http://www.turkstat.gov.tr/PreHaberBultenleri.do?id=15847, accessed 18 December 2017).

UNIATF (2017). UN Interagency Task Force on NCDs supports Turkey drive forward action on noncommunicable diseases. Ankara, 22 April 2016. Geneva: World Health Organization (http://www.who.int/ncds/un-task-force/turkey-mission-april-2016/en/, accessed 28 September 2017).

Ural D, Kayıkçıoğlu M (2014). Country report Turkey (update Nov 2014). Sophia Antipolis: European Society of Cardiology (https://www.escardio.org/Sub-specialty-communities/European-Association-of-Preventive-Cardiology-(EAPC)/Prevention-in-your-country/Country-of-the-Month-Turkey, accessed 6 December 2017).

Wang PS, Beck A, Berglund P, Leutzinger JA, Pronk N, Richling D et al. (2003). Chronic medical conditions and work performance in the health and work performance questionnaire calibration surveys. J Occup Environ Med. 45(12):1303–11.

WHO (2010). Global strategy to reduce the harmful use of alcohol. Geneva: World Health Organization (http:// www.who.int/substance_abuse/publications/global_strategy_reduce_harmful_use_alcohol/en/, accessed 23 August 2017).

WHO (2012). Costing tool – user guide. Scaling up action against noncommunicable diseases: how much will it cost? Geneva: World Health Organization (http://www.who.int/ncds/management/c_NCDs_costing_estimation_tool_user_manual.pdf, accessed 17 December 2017).

WHO (2013). Global action plan for the prevention and control of noncommunicable diseases 2013–2020. Geneva: World Health Organization (http://apps.who.int/medicinedocs/en/m/abstract/Js21446en/, accessed 21 August 2017).

WHO (2014a). Noncommunicable diseases country profiles 2014: Turkey. In: World Health Organization [website]. Geneva: World Health Organization (http://www.who.int/nmh/countries/en/#T, accessed 17 December 2017).

WHO (2016a). Cardiovascular risk prediction charts. In: World Health Organization [website]. Geneva: World Health Organization (http://www.who.int/cardiovascular_diseases/publications/Chart_predictions/en/, accessed 22 July 2017).

WHO (2016b). SHAKE the salt habit: the SHAKE technical package for salt reduction. Geneva: World Health Organization (http://www.who.int/dietphysicalactivity/publications/shake-salt-habit/en/, accessed 25 July 2017).

WHO (2016c). Diabetes country profiles 2016: Turkey. Geneva: World Health Organization (http://www.who.int/ diabetes/country-profiles/en/#T, accessed 6 December 2017).

WHO (2017a). Risk of premature death from the four target NCDs. In: Global Health Observatory data repository [online database]. Geneva: World Health Organization (http://apps.who.int/gho/data/node.main.A857?lang=en, accessed 2 December 2017).

WHO (2017b). Preparation for the third High-level meeting of the General Assembly on the Prevention and Control of Non-communicable Diseases to be held in 2018. Report by the Director-General. Geneva: World Health Organization (WHA A70/27; http://apps.who.int/gb/e/e_wha70.html, accessed 21 August 2017).

WHO (2017c). WHO Framework Convention on Tobacco Control [website]. Geneva: World Health Organization (http://www.who.int/fctc/en/, accessed 23 August 2017).

WHO (2017d). WHO report on the global tobacco epidemic, 2017: country profile – Turkey. Geneva: World Health Organization (http://www.who.int/tobacco/surveillance/policy/country_profile/en/#T, accessed 28 September 2017).

WHO (2017e). WHO report on the global tobacco epidemic, 2017. Geneva: World Health Organization (http://www.who.int/tobacco/global_report/2017/en/, accessed 1 September 2017).

WHO (2017f). Cost effectiveness and strategic planning (WHO-CHOICE). In: World Health Organization [website]. Geneva: World Health Organization (http://www.who.int/choice/cost-effectiveness/en/, accessed 28 December 2017).

WHO (2018). Global status report on alcohol and health, 2018. Geneva: World Health Organization (http://www. who.int/substance_abuse/publications/global_alcohol_report/en/, accessed 19 October 2018).

WHO Regional Office for Europe (2012). European action plan to reduce the harmful use of alcohol 2012–2020. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/health-topics/disease-prevention/ alcohol-use/publications/2012/european-action-plan-to-reduce-the-harmful-use-of-alcohol-20122021, accessed 12 December 2017).

WHO Regional Office for Europe (2014). Better noncommunicable disease outcomes: challenges and opportunities for health systems. Turkey country assessment. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/countries/turkey/publications/better-noncommunicable-disease-outcomes-challenges-and-opportunities-for-health-systems.-turkey-country-assessment-2014, accessed 16 December 2017).

WHO Regional Office for Europe (2015a). National healthy cities networks in the WHO European Region. Promoting health and well-being throughout Europe. Copenhagen: WHO Regional Office for Europe (http:// www.euro.who.int/en/health-topics/environment-and-health/urban-health/publications/2015/national-healthycities-networks-in-the-who-european-region.-promoting-health-and-well-being-throughout-europe-2015, accessed 6 December 2017).

WHO Regional Office for Europe (2015b). Ambulatory care sensitive conditions in Germany. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/countries/germany/publications/ambulatory-care-sensitive-conditions-in-germany-2015, accessed 17 December 2017)

WHO Regional Office for Europe (2015c). Ambulatory care sensitive conditions in Kazakhstan. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/countries/kazakhstan/publications/ambulatory-care-sensitive-conditions-in-kazakhstan-2015, accessed 17 December 2017).

WHO Regional Office for Europe (2015d). Ambulatory care sensitive conditions in Latvia. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/countries/latvia/publications/ambulatory-care-sensitive-conditions-in-latvia-2015, accessed 22 December 2017).

WHO Regional Office for Europe (2015e). Ambulatory care sensitive conditions in the Republic of Moldova. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/countries/republic-of-moldova/ publications/ambulatory-care-sensitive-conditions-in-the-republic-of-moldova-2015, accessed 17 December 2017).

WHO Regional Office for Europe (2016a). Turkish Healthy Nutrition and Active Life Programme 2010–2014 and related initiatives: an evaluation of progress since the WHO Ministerial Conference on Counteracting Obesity. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/countries/turkey/publications/turkish-healthy-nutrition-and-active-life-programme-20102014-and-related-initiatives-2016, accessed 19 December 2017).

WHO Regional Office for Europe (2016b). Ambulatory care sensitive conditions in Portugal. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/countries/portugal/publications/ambulatory-care-sensitive-conditions-in-portugal-2016, accessed 17 December 2017).

WHO Regional Office for Europe (2017a). Policy forum examines strengthening health financing in Turkey. In: WHO/Europe [website]. Copenhagen: WHO Regional Office for Europe (http://www.euro.who.int/en/countries/turkey/news/news/2017/06/policy-forum-examines-strengthening-health-financing-in-turkey, accessed 28 December 2017).

WHO Regional Office for Europe (2017b). European Health Information Gateway [online database]. Copenhagen: WHO Regional Office for Europe (https://gateway.euro.who.int/en/, accessed 25 August 2017).

WHO Regional Office for Europe (2017c). Monitoring noncommunicable disease commitments in Europe. Theme in focus: progress monitor indicators. Copenhagen: WHO Regional Office for Europe (http://www.euro.who. int/en/health-topics/noncommunicable-diseases/pages/who-european-office-for-the-prevention-and-control-of-noncommunicable-diseases-ncd-office/publications-and-tools/monitoring-noncommunicable-disease-commitments-in-europe.-theme-in-focus-progress-monitor-indicators-2017, accessed 26 October 2017).

Annex 1. Contributors to the report

İrfan ŞENCAN, Former Director-General of Public Health, Ministry of Health Bekir KESKİNKILIÇ, Directorate-General of Public Health, Ministry of Health Pavel URSU, World Health Organization Jill FARRINGTON, World Health Organization Toker ERGÜDER, World Health Organization Banu EKİNCİ, Directorate-General of Public Health, Ministry of Health Anna KONTSEVAYA, World Health Organization, Consultant Mehmet BALCILAR, World Health Organization, Consultant Yavuz ATEŞ, Directorate-General of Public Health, Ministry of Health Nazan YARDIM, Directorate-General of Public Health, Ministry of Health Zübeyde ÖZKAN ALTUNAY, Directorate-General of Public Health, Ministry of Health Gülay SARIOĞLU, Directorate-General of Public Health, Ministry of Health Asiye KAPUSUZ NAVRUZ, Directorate-General of Public Health, Ministry of Health Ahmet ARIKAN, Directorate-General of Public Health, Ministry of Health Fatma GÜRBÜZTÜRK, Directorate-General of Public Health, Ministry of Health Nevin COBANOGLU, Directorate-General of Public Health, Ministry of Health Seçil SİS, Directorate-General of Public Health, Ministry of Health Sibel GÖGEN, Directorate-General of Public Health, Ministry of Health Seniz ILGAZ, Directorate-General of Public Health, Ministry of Health Aslı SUNGUR, Directorate-General of Public Health, Ministry of Health Peyman ALTAN, Directorate-General of Public Health, Ministry of Health Türkan ALGIN, Directorate-General of Public Health, Ministry of Health Esra TİRYAKİOĞLU, Directorate-General of Public Health, Ministry of Health İlayda ÜÇÜNCÜ, Directorate-General of Public Health, Ministry of Health Neriman İLHAN, Directorate-General of Public Health, Ministry of Health Özgür AYDOĞ, Directorate-General of Public Health, Ministry of Health Esra GÜNEŞ, Directorate-General of Public Health, Ministry of Health Tolga OZAN, Directorate-General of Public Health, Ministry of Health Ilhami AKKUŞ, Directorate-General of Public Health, Ministry of Health Berrak BAŞARA, Directorate-General of Health Research, Ministry of Health Ayfer PEKERİÇLİ, Directorate-General of Health Research, Ministry of Health Tuncay KOYUNCU, Directorate-General of Health Services, Ministry of Health Hatice AKTAŞ, Directorate-General of Health Information Systems, Ministry of Health Ali Kemal ÇAYLAN, Directorate-General of Public Hospitals, Ministry of Health Sevda POLAT, Directorate-General of Public Hospitals, Ministry of Health Esra ŞAFAK YILMAZ, Turkish Medicine and Medical Devices Agency, Ministry of Health Melda KEÇİK, Turkish Medicine and Medical Devices Agency, Ministry of Health Aslihan BEYAN, Turkish Medicine and Medical Devices Agency, Ministry of Health Erkan GÜLBEYAZ, Social Security Institution, Ministry of Labour and Social Policies Esin ÜNSA, Social Security Institution, Ministry of Labour and Social Policies Dilek YILMAZ, Social Security Institution, Ministry of Labour and Social Policies Deniz YİĞİT, Social Security Institution, Ministry of Labour and Social Policies Sila Saadet TOKER, World Health Organization Gül MENET, World Health Organization

Annex 2. Data sources used for the economic burden analysis

- Population by age and sex for the period 2015–2031, projected based on the intermediate growth scenario of the Turkish Statistical Institute
- Mortality statistics by age and sex groups and causes of death (2013–2015)
- Registered morbidity statistics by disease groups (2013–2015)
- Economic data (GDP, employment and unemployment rates and other)
- Epidemiology data on NCDs, prevalence of risk factors and incidence in age and sex groups (mainly based on a chronic diseases and risk factors survey undertaken in Ankara, Turkey, in 2013)

Annex 3. Population projections: detailed methodology and results

The cohort component method was used for population projection. This takes the components of demographic change and projects population figures in a way that reproduces how populations grow or decline over time. The components of change include mortality, fertility and net migration. Alternative methods such as the application of constant population growth yield much less satisfactory results than the cohort component method.

In this population projection, to calculate the mortality component, each "cohort" or age group is carried forward in time, assuming that it is subject to the standard mortality rate this age group is exposed to. Thus, the mortality rate used as the age group is carried forward is the rate that group is exposed to under demographic change characteristics. For example, the group aged 10–14 years is carried forward subject to relatively low mortality rates, whereas the 70–74-year age group is naturally subject to much higher rates. These calculations are performed on a by-sex basis, as there may be significant variations in mortality by sex.

In order to calculate the probability of death, the Coale-Demeny-West model life tables were used,¹ which are considered the most suitable for the mortality rates in Turkey, adjusted for the significant improvement in Turkey's mortality level in recent years. The updated model life tables were obtained from the Turkish Statistical Institute (TURKSTAT) for each age from 0 to 99 years and for 100 years and above as the highest age group.

The second component of the cohort component method is fertility, or the number of births among women of reproductive age at an assumed birth rate. Projected births fill the youngest age group in each period of time and, in turn, are subject to infant mortality rates.

The third and final component is migration by age and gender in each period. Migration can be either positive net immigration or net emigration. Migration is almost always the component of the project most difficult to estimate. Using 2010, 2011 and 2012 data on international migration, the residual method was used to obtain annual net migration magnitudes by sex and age. This estimates net migration as the net population change between two time points minus the difference between births and deaths between the points. Thus, the residual method does not require any statistics on migrants or emigrants and estimates net migration indirectly.

Population projections by age and sex used in the analysis are given in Table A3 for the period 2016 to 2031.

¹ Coale AJ, Demeny P (1983). Regional model life tables and stable populations, 2nd edition. New York: Academic Press.

ons
jecti
pro
ation
pula
. Po
le A3
Tabl

Population by sex and age (0-4 years, 5-9 years, 10-14 years, ... 75-79 years, 80+ years) for 2016 and projecting upwards to 2031, inclusive

רטטטומוטוו אל אבא מווע מצב נעדע אבמוא, אדש אבמוא, בעדע אבמוא	y sex allu	dgc / 1		1 Yuuu, H	577 FT 0)		an hand in	1 101 2010		and projecting apwards	3				
Gender and age (years)	2016	2017	2018	2019	2020	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030	2031
Females 0–4	3 305 566	3 261 383	3 187 798	3 109 143	3 049 342	2 979 369	2 949 523	2 951 542	2 960 846	2 955 388	2 947 606	2 931 074	2 905 558	2 877 997	2 856 016	2 837 538
Females 5–9	3 241 101	3 247 496	3 261 310	3 273 411	3 270 858	3 249 402	3 211 190	3 159 567	3 105 868	3 063 278	3 033 380	3 006 773	2 985 287	2 965 759	2 943 722	2 920 136
Females 10–14	3 250 027	3 238 569	3 228 528	3 221 642	3 219 969	3 223 399	3 228 234	3 230 919	3 223 697	3 203 435	3 181 719	3 151 107	3 116 486	3 082 776	3 052 600	3 024 894
Females 15–19	3 273 830	3 272 294	3 265 283	3 253 500	3 241 921	3 231 400	3 215 201	3 198 760	3 186 435	3 176 210	3 174 655	3 178 358	3 181 077	3 177 599	3 163 493	3 147 784
Females 20–24	3 134 982	3 165 168	3 201 706	3 234 584	3 252 897	3 264 404	3 260 316	3 244 988	3 225 711	3 206 460	3 191 810	3 176 340	3 162 910	3 152 380	3 143 331	3 142 747
Females 25–29	3 133 990	3 118 549	3 107 334	3 104 165	3 111 207	3 132 388	3 163 068	3 195 746	3 221 683	3 233 685	3 236 211	3 227 815	3 212 363	3 193 739	3 175 591	3 161 886
Females 30–34	3 180 603	3 176 079	3 166 937	3 154 939	3 141 141	3 131 388	3 118 955	3 106 305	3 101 839	3 106 636	3 120 163	3 144 041	3 173 003	3 195 757	3 206 843	3 209 228
Females 35–39	3 059 608	3 095 735	3 128 195	3 151 952	3 166 087	3 175 393	3 173 093	3 162 582	3 149 172	3 134 869	3 119 154	3 102 659	3 088 228	3 081 767	3 083 852	3 096 412
Females 40–44	2 802 739	2 858 670	2 911 635	2 960 804	3 005 438	3 050 378	3 088 878	3 120 374	3 143 130	3 156 044	3 158 509	3 152 116	3 140 708	3 125 144	3 108 047	3 092 383
Females 45–49	2 445 703	2 512 495	2 584 809	2 657 158	2 723 054	2 788 345	2 846 260	2 898 280	2 945 740	2 989 671	3 028 334	3 063 295	3 093 274	3 114 047	3 124 177	3 126 631
Females 50–54	2 174 950	2 213 932	2 256 989	2 305 724	2 360 845	2 424 300	2 492 357	2 563 631	2 633 542	2 698 266	2 757 893	2 812 983	2 863 171	2 909 269	2 951 787	2 989 639
Females 55–59	1 958 744	1 994 721	2 030 495	2 066 789	2 104 405	2 143 265	2 182 567	2 223 958	2 270 989	2 325 188	2 383 514	2 449 627	2 519 025	2 587 474	2 651 366	2 709 612
Females 60–64	1 659 230	1 718 971	1 774 200	1 823 871	1 869 917	1 910 236	1 946 966	1 981 764	2 016 195	2 052 941	2 087 845	2 125 634	2 166 806	2 213 223	2 266 263	2 323 820
Females 65–69	1 203 016	1 282 533	1 369 889	1 455 513	1 532 654	1 599 198	1 656 224	1 706 408	1 752 338	1 794 811	1 830 523	1 865 229	1 900 370	1 935 814	1 973 907	2 007 530
Females 70–74	898 543	925 447	958 624	1 002 532	1 059 686	1 124 139	1 201 063	1 284 328	1 366 622	1 440 890	1 496 508	1 545 274	1 590 538	1 635 203	1 678 527	1 710 380
Females 75–79	655 559	675 487	700 342	729 748	760 340	784 097	809 064	839 134	880 197	936 730	991 954	1 060 799	1 135 378	1 209 505	1 277 293	1 323 580
Females 80+	764 654	793 524	819 549	843 242	867 107	906 112	944 409	979 828	1 015 147	1 048 654	1 094 883	1 140 535	1 188 867	1 241 786	1 304 513	1 386 032
Males 00–04	3 469 208	3 423 064	3 341 774	3 255 491	3 190 035	3 113 385	3 081 861	3 086 206	3 098 818	3 096 553	3 089 890	3 072 379	3 045 840	3 018 215	2 995 137	2 975 537
Males 05–09	3 378 957	3 388 347	3 406 345	3 422 746	3 423 525	3 401 421	3 360 571	3 305 285	3 246 860	3 202 427	3 169 609	3 143 032	3 122 542	3 102 951	3 081 835	3 058 135
Males 10–14	3 381 933	3 369 501	3 359 655	3 352 061	3 352 680	3 358 416	3 366 587	3 370 607	3 365 697	3 345 608	3 321 985	3 290 393	3 254 750	3 219 967	3 189 705	3 160 878
Males 15–19	3 391 850	3 393 306	3 388 464	3 378 941	3 368 645	3 357 416	3 341 523	3 324 379	3 312 321	3 303 259	3 302 812	3 308 561	3 314 294	3 311 765	3 299 590	3 283 768
Males 20–24	3 216 307	3 252 456	3 296 078	3 334 141	3 357 669	3 372 417	3 370 597	3 357 543	3 340 520	3 322 417	3 307 857	3 292 412	3 278 971	3 269 397	3 263 298	3 263 622
Males 25–29	3 173 661	3 163 185	3 157 003	3 160 913	3 173 072	3 201 396	3 238 259	3 276 142	3 307 286	3 323 425	3 330 058	3 324 710	3 311 267	3 295 624	3 279 428	3 266 644
Males 30–34	3 176 636	3 177 071	3 173 891	3 166 886	3 159 102	3 155 391	3 148 029	3 142 483	3 144 137	3 155 036	3 173 646	3 204 601	3 239 612	3 267 379	3 283 460	3 289 812
Males 35–39	3 008 036	3 051 099	3 089 453	3 119 099	3 138 148	3 152 390	3 156 050	3 151 528	3 144 137	3 135 878	3 125 209	3 115 780	3 108 412	3 109 004	3 118 128	3 136 703
Males 40–44	2 678 768	2 743 609	2 809 316	2 871 204	2 925 612	2 978 369	3 023 712	3 061 082	3 088 747	3 107 645	3 116 127	3 115 780	3 110 431	3 101 942	3 092 925	3 083 317
Males 45–49	2 312 805	2 373 628	2 439 773	2 506 828	2 572 383	2 641 327	2 708 909	2 772 661	2 831 939	2 886 822	2 933 478	2 975 484	3 011 527	3 038 390	3 056 632	3 065 186
Males 50–54	2 005 357	2 051 259	2 100 033	2 152 407	2 207 181	2 267 281	2 328 940	2 391 784	2 457 301	2 522 819	2 586 345	2 650 482	2 712 796	2 772 077	2 825 772	2 871 786
Males 55–59	1 699 892	1 749 720	1 799 035	1 848 761	1 897 856	1 945 241	1 991 078	2 038 041	2 088 706	2 142 682	2 196 829	2 256 846	2 319 199	2 384 713	2 449 742	2 511 176
Males 60–64	1 342 855	1 400 570	1 457 308	1 514 251	1 568 575	1 617 200	1 665 247	1 713 442	1 761 402	1 808 927	1 852 723	1 897 527	1 943 767	1 994 322	2 048 508	2 102 216
Males 65–69	954 082	1 005 792	1 065 911	1 127 973	1 189 403	1 244 154	1 297 309	1 348 645	1 399 856	1 449 965	1 493 480	1 539 218	1 586 501	1 636 211	1 685 583	1 727 504
Males 70–74	701 181	713 180	731 137	758 619	796 261	836 103	884 255	939 630	997 020	1 053 695	1 096 901	1 141 544	1 186 848	1 233 716	1 283 342	1 323 580
Males 75–79	507 786	510 831	516 564	525 657	538 823	549 068	560 4 29	575 837	600 226	634 234	668 030	708 545	754 900	801 966	849 849	887 423
Males 80+	423 485	448 341	465 901	476875	480 950	508 063	529350	543 679	551 886	554576	583 265	605 594	624 710	643 590	663 346	712 155

Annex 4. Data used for calculations of NCD burden

This annex includes the figures from different literature sources used in the NCD burden calculations, including the details of the share of government health care expenditure on NCDs per country (Table A4.1) and of the reduction in productivity by disease (Table A 4.2).

Category	۶	Australia	Canada	Czechia	Germany	Estonia	France	Georgia	Hungary	India	Republic of Korea	The Netherlands	Slovenia	United States of America
World E group	World Bank income group	High	High	High	High	High	High	Low- middle	High	Low- middle	High	High	High	High
nre	CVD	8.6%	9.0%	9.2%	16.2%	22.0%	12.0%	0.8%	18.4%	15.6%	13.4%	11.1%	13.6%	17.0%
	Cancers (neoplasm)	4.7%	3.1%	5.5%	7.9%	9.4%	7.1%	1.9%	8.2%	4.7%	7.7%	5.5%	6.7%	7.2%
ə dfleəd fo noif by disease gro	Endocrine and metabolic diseases	4.4%	2.1%	2.0%	4.0%	5.4%	3.9%	%0.0	6.1%	4.8%	4.1%	2.7%	2.4%	4.4%
Propor	Respiratory diseases	4.8%	4.5%	3.2%	5.4%	7.2%	7.3%	0.4%	5.8%	8.9%	10.8%	4.9%	6.2%	6.6%
Total fo	Total for four NCDs	22.5%	18.7%	19.9%	33.5%	44.0%	30.3%	3.1%	38.5%	34.0%	36.0%	24.2%	28.9%	35.2%

Table A4.1. Share of major NCDs in total health care expenditure

Source: Garg CC, Evans DB (2011). What is the impact of noncommunicable diseases on national health expenditures: a synthesis of available data. Geneva: World Health Organization (NCD discussion paper no. 3; http:// www.who.int/healthsystems/NCDdiscussionpaper3.pdf, accessed 23 August 2017).

Table A4.2. Productivity data	oductivity	data		
Disease	Parameter value	Year	Source of data	Details of data source
Labour force participation rate reduction	ticipation rate	e reduction		
Hypertension	11.0%	2009	Institute for Research and Information in Health Economics	Barnay T, Debrand T (2006). Effects of health on the labour force participation of older persons in Europe. Questions d'économie de la Santé. 109 (http://www.irdes. fr/english/2006/issues-in-health-economics.html, accessed 8 September 2017).
Stroke	11.0%	2009	Institute for Research and Information in Health Economics	Barnay T, Debrand T (2006). Effects of health on the labour force participation of older persons in Europe. Questions d'économie de la Santé. 109 (http://www.irdes. fr/english/2006/issues-in-health-economics.html, accessed 8 September 2017).
AMI	11.0%	2009	Institute for Research and Information in Health Economics	Barnay T, Debrand T (2006). Effects of health on the labour force participation of older persons in Europe. Questions d'économie de la Santé. 109 (http://www.irdes. fr/english/2006/issues-in-health-economics.html, accessed 8 September 2017).
Type 2 diabetes	11.0%	2009	Institute for Research and Information in Health Economics	Barnay T, Debrand T (2006). Effects of health on the labour force participation of older persons in Europe. Questions d'économie de la Santé. 109 (http://www.irdes. fr/english/2006/issues-in-health-economics.html, accessed 8 September 2017).
Reduction in full-time hours due to absenteeism	time hours du	ue to abse	nteeism	
Hypertension	0.5%	2011	Population Health Management	Mitchell RJ, Bates P (2011). Measuring health-related productivity loss. Popul Health Manag. 14(2): 93–8.
Stroke	5.5%	2011	Population Health Management	Mitchell RJ, Bates P (2011). Measuring health-related productivity loss. Popul Health Manag. 14(2): 93–8.
AMI	1.1%	2011	Population Health Management	Mitchell RJ, Bates P (2011). Measuring health-related productivity loss. Popul Health Manag. 14(2): 93–8.
Type 2 diabetes	0.3%	2011	Population Health Management	Mitchell RJ, Bates P (2011). Measuring health-related productivity loss. Popul Health Manag. 14(2): 93–8.
Reduction in productivity due to presenteeism	ductivity due	to present	teeism	
CVD/stroke	3.7%	2011	Wang PS, Beck A, Berglund P, Le work performance in the health 45(12):1303–11.	Wang PS, Beck A, Berglund P, Leutzinger JA, Pronk N, Richling D et al. (2003). Chronic medical conditions and work performance in the health and work performance questionnaire calibration surveys. J Occup Environ Med. 45(12):1303–11.

τ	5
£.	5
Ś	
C	5
č	5
C	1
5	ţ
<	ζ
0	D
2	2
ĥ	0

Disease	Parameter value	Year	Source of data	Details of data source
Diabetes	11%	2011	Holden L, Scuffham productivity in work	Holden L, Scuffham PA, Hilton MF, Ware RS, Vecchio N, Whiteford HA (2011). Which health conditions impact on productivity in working Australians? J Occup Environ Med.53(3):253–7. doi:10.1097/JOM.0b013e31820d1007.
Time it takes to replace workers (percentage of a year)	20%	2012	2012 Center for American Progress	Boushey H, Glynn SJ (2012). There are significant business costs to replacing employees. Washington, DC: Center for American Progress (https://www.americanprogress.org/issues/labor/report/2012/11/16/44464/there-are-significant-business-costs-to-replacing-employees/, accessed 8 September 2017).

The WHO Regional Office for Europe

The World Health Organization (WHO) is a specialized agency of the United Nations created in 1948 with the primary responsibility for international health matters and public health. The WHO Regional Office for Europe is one of six regional offices throughout the world, each with its own programme geared to the particular health conditions of the countries it serves.

Member States

Albania Andorra Armenia Austria Azerbaijan Belarus Belgium Bosnia and Herzegovina Bulgaria Croatia Cyprus Czechia Denmark Estonia Finland France Georgia Germany Greece Hungary Iceland Ireland Israel Italy Kazakhstan Kyrgyzstan Latvia Lithuania Luxembourg Malta Monaco Montenegro Netherlands Norway Poland Portugal **Republic of Moldova** Romania **Russian Federation** San Marino Serbia Slovakia Slovenia Spain Sweden Switzerland Tajikistan The former Yugoslav Republic of Macedonia Turkey Turkmenistan Ukraine United Kingdom Uzbekistan

World Health Organization Regional Office for Europe

UN City, Marmorvej 51, DK-2100 Copenhagen Ø, Denmark Tel.: +45 45 33 70 00 Fax: +45 45 33 70 01 Email: euwhocontact@who.int Website: www.euro.who.int